

# Central Lancashire Online Knowledge (CLoK)

Title	Forecasting economic activity using preselected predictors: the case of
	Cyprus
Туре	Article
URL	https://clok.uclan.ac.uk/id/eprint/45657/
DOI	
Date	2022
Citation	Anaxagorou, Christiana and Pashourtidou, Nicoletta (2022) Forecasting
	economic activity using preselected predictors: the case of Cyprus. Cyprus
	Economic Policy Review, 16 (2). pp. 11-36.
Creators	Anaxagorou, Christiana and Pashourtidou, Nicoletta

It is advisable to refer to the publisher's version if you intend to cite from the work.

For information about Research at UCLan please go to <a href="http://www.uclan.ac.uk/research/">http://www.uclan.ac.uk/research/</a>

All outputs in CLoK are protected by Intellectual Property Rights law, including Copyright law. Copyright, IPR and Moral Rights for the works on this site are retained by the individual authors and/or other copyright owners. Terms and conditions for use of this material are defined in the <u>http://clok.uclan.ac.uk/policies/</u>

### Forecasting economic activity using preselected predictors: the case of Cyprus

Christiana Anaxagorou and Nicoletta Pashourtidou\*

Economics Research Centre, University of Cyprus

#### Abstract

This paper applies hard and soft thresholding techniques to a large dataset of domestic and foreign series in order to preselect informative predictors for forecasting economic activity in Cyprus. The variables of interest in the forecasting exercise consist of GDP and the production-side components of GDP, expressed in growth rates. The subsets of selected predictors are allowed to differ across the variables of interest and over the forecast horizon, thus accommodating idiosyncratic features of economic sectors. The sets of selected predictors contain a higher proportion of domestic as opposed to foreign predictors for the one-quarter horizon, while the opposite occurs for longer horizons. Furthermore, in the case of GDP all thresholding techniques result in selecting high proportions of business and consumer survey indicators for all horizons. The forecasting performance depends on the forecast horizon and, most importantly, on whether the subsets of chosen predictors remain constant or change over time. The thresholding technique employed is not found to substantially affect the forecasting performance. Selecting predictors prior to forecasting GDP growth, leads to lower forecast errors vis-à-vis a simple univariate benchmark, as well as compared to exploiting the full dataset of predictors for forecasting. The gains from preselecting predictors are higher during a crisis period than in normal times, especially for short horizons, while preselection in normal times benefits forecast accuracy for longer horizons. Predictor preselection is found to improve the forecasting performance in the case of some production-side components, particularly the gross value added in the sectors of trade and construction, and net taxes.

Keywords: Forecasting, factor models, predictor selection

#### 1. Introduction

Effective economic decisions by policy makers or private agents are dependent on the available information. Economic agents and policy makers, however, are likely to have imperfect information on various indicators of economic performance at the aggregate and sectoral levels, because of delays in the publication of official data, particularly national accounts, highlighting the importance of forecasting economic activity in economic and financial decision-making. The quantity of available data has increased rapidly in recent years, mainly because of technological improvements and the rise of big data relating to economic and financial indicators. The availability of many

<sup>\*</sup> Corresponding Author Address: Economics Research Centre, University of Cyprus, P.O. Box 20537, 1678, Nicosia, Cyprus. Email: <u>n.pashourtidou@ucy.ac.cy</u>.

candidate predictors has given rise to forecasting methods that exploit the richness of information without proportionally increasing the number of estimated parameters. Such methods, which include forecast combinations (or forecast pooling), dynamic factor models and Bayesian model averaging, have been widely applied and found to improve upon the forecasting performance of univariate benchmarks for many macroeconomic variables (see Stock & Watson, 2006 for a survey).

The various indicators that can be employed to forecast activity developments are released according to different publication schedules and with different frequencies, resulting in ragged-edge data and mixed frequency problems that have been extensively studied in the literature (see e.g. Stock & Watson, 2002; Kitchen & Monaco, 2003; Baffigi, Golinelli & Parigi, 2004; Giannone, Reichlin & Small, 2008; Banbura & Runstler, 2011; Foroni & Marcellino, 2014). Typically, an economic activity variable in quarterly frequency such as GDP, consumption, or sectoral value added, is forecast using indicators published on a monthly basis, stressing the need for timely available monthly data. While the plethora of available data, along with dimension reduction methods, can assist practitioners and policy makers in their efforts to forecast economic activity, the use of many noisy indicators (usually of higher publication frequency than the variable to be forecast) may not necessarily improve the forecasting performance. There is a growing body of literature that investigates the effects of indicator preselection from large datasets on the accuracy of forecasts for macroeconomic variables (see e.g. Boivin & Ng, 2006; Bai & Ng, 2008; Kim & Swanson, 2014, 2018; Jokubaitis, Celov & Leipus, 2021).

Prior to forecasting, indicators can be selected from a large dataset by applying "hard" thresholding rules, i.e. a statistical significance test, or "soft" thresholding methods, i.e. optimisation algorithms that carry out variable selection and shrinkage at the same time, providing also a ranking of the predictors (see e.g. Bai & Ng, 2008). The information in selected predictors is usually summarized through a small number of estimated common factors, which are subsequently used in forecasting models. The simulations in Boivin and Ng (2006) show that factors extracted from larger panels of series could lead to higher forecast errors compared to factors estimated from smaller datasets; this occurs when the extended panel includes series with cross-correlated idiosyncratic errors and/or higher error variances than those in the smaller panel. The Monte Carlo study in Boivin and Ng (2006) also demonstrates that the forecasting power of factors estimated from a given panel depends on their dominance in the panel (i.e. the size of factor loadings) and, therefore, on how informative the dataset is about the factors. A factor that is dominant in a smaller dataset can lead to inferior forecasts, if it is dominated by another factor in a larger dataset. The empirical analysis in Boivin and Ng (2006) that includes many economic variables reveals that factors extracted from the full set of 147 series do not yield superior forecasts vis-à-vis factors estimated from smaller sets of prescreened indicators.

Empirical studies on forecasting GDP growth in the euro area as a whole and in large euro area economies, have documented gains associated with applying preselection techniques to large datasets vis-à-vis utilising the full set of predictors (see e.g. Schumacher, 2010; Caggiano, Kapetanios & Labhard, 2011; Bulligan, Marcellino & Venditti, 2015; Girardi, Golinelli & Pappalardo, 2017). Moreover, the empirical results of Caggiano, Kapetanios and Labhard (2011) highlight the benefits of combinations of factor-based forecasts obtained from prescreened datasets.

Empirical evidence, particularly for the euro area, provides the motivation for considering subsets of predictors drawn from large datasets in forecasting economic

activity in Cyprus. Previous work using data for Cyprus has shown that GDP and sectoral growth forecasts computed from a set of preselected predictors, using a simple hard thresholding method (i.e. significant correlations with dependent variables), are at least as accurate as the forecasts obtained using the full set of predictors (Pashourtidou, Papamichael & Karagiannakis, 2018). This paper builds on previous work on the performance of methods for forecasting activity growth in Cyprus by applying hard and soft thresholding techniques to a large dataset of domestic and foreign series in order to preselect informative predictors. The variables of interest in the forecasting exercise are GDP and its production-side components, expressed in growth rates. The subsets of selected predictors are allowed to differ across the variables of interest and over the forecast horizon, thus addressing the idiosyncratic features of sectors. Our results show that the forecasting performance depends on the forecast horizon and on whether the subsets of chosen predictors remain constant or change over time. The forecasting performance is not found to vary considerably with the thresholding technique employed for selecting the predictors. Selecting a subset of predictors before forecasting GDP growth, leads to lower forecast errors vis-à-vis a simple univariate benchmark, as well as compared to exploiting the full dataset of predictors for forecasting; the gains from preselecting predictors are higher during a crisis period than in normal times. Predictor preselection is found to improve the forecasting performance in the case of some production-side components, particularly the gross value added in the sectors of trade and construction, and net taxes.

The structure of the paper is as follows. Section 2 presents the forecasting model when a large number of predictors is used and gives an overview of predictor selection methods. Section 3 briefly describes the dataset and details the thresholding techniques and schemes employed for preselecting predictors from the full dataset. Section 4 describes the forecasting exercise and presents the results; it also discusses the forecasting performance during crisis and normal periods. Section 5 concludes.

#### 2. Forecasting with preselected predictions

We are interested in forecasting a variable,  $y_t$ , using data on a large number of predictors,  $X_t = (X_{1,t}, ..., X_{N,t})'$ , and t = 1, ..., T. In many empirical applications the number of available predictors, N, is larger than the number of available time periods, T. The factor approach is used to solve the dimensionality problem and obtain feasible forecasts for  $y_t$ , while utilising the information in  $X_t$ . In the factor approach, the h-step ahead forecast,  $y_{t+h}^h$ , for  $y_t$  is given by the model

$$y_{t+h}^{h} = \alpha_{h} + \sum_{j=0}^{p} \beta_{h,j}' F_{t-j} + \sum_{m=0}^{q} \gamma_{h,m} y_{t-m} + \varepsilon_{t+h},$$
(1)

and the series of predictors have a factor representation

$$X_{i,t} = \lambda'_i F_t + e_{i,t}, \qquad i = 1, ..., N$$
 (2)

where  $F_t$  is an  $r \times 1$  vector of factors, i.e. a component common to all predictors,  $\lambda_i$  is a vector of factor loadings for series *i* and  $e_{it}$  is the idiosyncratic error. As the vector of factors is unobserved, an estimate,  $\hat{F}_t$ , is obtained using principal components.

Equation (1) is then estimated with data up to period T - h and by replacing  $F_t$  with  $\hat{F}_t$ .<sup>1</sup> The *h*-step ahead forecast is computed as

$$\hat{y}_{T+h}^{h} = \hat{\alpha}_{h} + \sum_{j=0}^{p} \hat{\beta}_{h,j}' \hat{F}_{T-j} + \sum_{m=0}^{q} \hat{\gamma}_{h,m} y_{T-m}.$$
(3)

The number of factors is typically estimated using information criteria based solely on the variation in the panel of predictors,  $X_t$ , without involving the variable to be forecast. However, estimated factors that capture a high proportion of the variation in the panel may not necessarily have a strong predictive content for the variable of interest. An issue investigated in the literature is how the quantity of information, i.e. the size of N, affects the accuracy of the forecasts in (3) through the quality of information conveyed by the estimated factors in  $\hat{F}_t$ . The effects of the number of predictors (N) on the forecasting performance of models similar to (1) in terms of mean squared forecast error have been widely investigated in the literature. Applications cover various large countries and a range of variables. For example, Bai and Ng (2008) focus on US inflation; Caggiano, Kapetanios and Labhard (2011) look at GDP growth for euro area countries and the UK; Kim and Swanson, (2014, 2018) consider key macroeconomic variables for the US; Bulligan, Marcellino and Venditti (2015) examine GDP growth for Italy with reference to other major euro area member states and the euro area as a whole; Panagiotelis et al. (2019) study main macroeconomic indicators for Australia. The above empirical applications use hard and/or soft thresholding techniques for selecting the predictors employed in the estimation of factors, which are subsequently included in forecasting equations.

Under a hard thresholding method, a predictor is chosen based on its statistical significance for the variable of interest; for example, a test for the significance of the correlation coefficient between a predictor and the variable of interest can be applied. Hard thresholding methods ignore information in other candidate predictors and may result in selecting collinear predictors, i.e. indicators whose information content is very similar. Another limitation of hard thresholding is that the set of selected predictors depends on the choice of the test or the decision rules used.

Soft thresholding methods are algorithms that select predictors and perform shrinkage, i.e. reduce the size of the estimates associated with predictors that are not important for the variable of interest. A widely used method, known as the Least Absolute Shrinkage Selection Operator (LASSO), performs a penalised regression on all available predictors with a penalty that results in some coefficient estimates being exactly equal to zero. In other words, LASSO selects a subset of predictors by shrinking to zero the coefficient estimates of uninformative regressors and therefore excluding the latter from the active set of variables. Another popular model selection algorithm is the forward selection method. The algorithm starts by adding to the null model the predictor with the maximum correlation with the variable of interest; at each step the predictor with the maximum correlation with the residual from the previous step is selected, until the model cannot be improved further (based on a statistical criterion) by including additional regressors. A shortcoming of the forward selection algorithm is that it excludes useful predictors that are correlated with the regressors selected in previous steps. A more general method, the Least Angle Regression (LARS) is a less aggressive selection algorithm than the forward selection method (see Efron et al.,

<sup>&</sup>lt;sup>1</sup>Bai and Ng (2006) show that estimating (1) by least squares when  $F_t$  is replaced by the principal component estimate,  $\hat{F}_t$ , leads to consistent and asymptotically normal parameter estimates and forecasts; hence, the estimated factors in  $\hat{F}_t$  can be treated as observed variables.

2004). More specifically, LASSO and a more cautious variant of the forward selection algorithm, known as the forward stagewise regression, can be implemented through modifications of the LARS method. Bai and Ng (2008) list the advantages of LARS. For example, LARS utilises the information in all candidate predictors and ranks them in order of importance with respect to the variable of interest; it excludes highly correlated predictors without being as aggressive as the forward selection method; and it is fast to compute.

In the empirical analysis that follows, we apply hard and soft thresholding methods to a large dataset of macroeconomic and financial indicators in order to select a subset of predictors that are informative for economic activity growth in Cyprus.

#### 3. Data and predictor selection

As we focus on forecasting economic activity at the aggregate and sectoral levels, the variables of interest are GDP and the production-side components of GDP obtained from quarterly national accounts.<sup>2</sup> In the analysis that follows, the production-side components of GDP are given by the Gross Value Added (GVA) in 10 sectors of economic activity and net taxes on products, as shown in Table A1. The predictors consist of 200 indicators that cover domestic and external economic and financial conditions, and are published well before the release of the quarterly national accounts. The dataset includes indicators for real activity and labour market (e.g. volume indices of retail trade and manufacturing, building permits, vehicle registrations, unemployment), financial series (e.g. stock market indices, interest rates, exchange rates, loans), price indices and international commodity prices, as well as series from business and consumer surveys in Cyprus and the EU; the full list of indicators is given in Table A2.<sup>3</sup> Over 90% of indicators in the dataset are published on a monthly basis, while a small number of quarterly series with forward-looking properties are also included.<sup>4</sup> Table 1 summarises the composition of the dataset; 60% of variables are domestic indicators and the remaining 40% represents foreign or international series. The two largest groups of indicators consist of survey and financial indicators, which are usually available at the end of the reference month. The dataset spans the period from 1995Q1 to 2019Q4.

<sup>&</sup>lt;sup>2</sup> National accounts data were obtained from the Statistical Service of Cyprus; the seasonally adjusted chain-linked volume measures are used.

<sup>&</sup>lt;sup>3</sup> The data were obtained from various local and international sources. Local sources include the Statistical Service of Cyprus, the Central Bank of Cyprus, the Cyprus Stock Exchange, the Department of Lands and Surveys, and the Department of Registrar of Companies and Intellectual Property. The sources of foreign/international variables include: Eurostat, the European Central Bank, the European Commission, Datastream and Global Financial Data. All data are seasonally adjusted and transformed into stationary series by differencing the levels of the series, or the logarithm of the series levels.

<sup>&</sup>lt;sup>4</sup> Quarterly series with forward-looking properties relate to information from the Survey of Professional Forecasters (European Central Bank) and quarterly questions in Business and Consumer Surveys in Cyprus (European Commission).

1	
Prodictors	Number of variables
riediciois	(percentage)
Total	200 (100%)
Domestic	117 (60%)
Foreign/international	83 (40%)
By category	
Real activity and labour market	45 (23%)
Financial	66 (33%)
Prices	18 (9%)
Business and consumer surveys	71 (36%)

TABLE 1 Composition of the dataset

In the empirical analysis, we employ hard and soft thresholding algorithms along the lines of other studies (e.g. Bai & Ng, 2008; Bulligan, Marcellino & Venditti, 2015; Girardi, Golinelli & Pappalardo, 2017) to select indicators from the panel of 200 series; the smaller set of selected indicators is then used for forecasting the growth rate of GDP and its production-side components. We apply a hard thresholding algorithm based on an F-test, as well as the LARS and LASSO methods.

As a hard thresholding rule, we consider the statistical significance of a candidate predictor after controlling for lagged values of the dependent variable. More specifically, for each candidate predictor,  $X_{i,t}$ , a regression of the variable of interest,  $y_t^h$ , on  $Z_{t-h}$  and  $X_{i,t-h}$  is run, where  $Z_{t-h}$  is a vector of deterministic terms and lags of  $y_t$ , and  $X_{i,t-h}$  is a vector of lags and monthly leads (if available) of the candidate predictor.<sup>5</sup> Next, the p-value,  $pv_{[i]}$ , corresponding to the F-statistic for testing the statistical significance of  $X_{i,t-h}$ , is obtained. The p-values from the *N* regressions are ranked in ascending order. Predictor *i* is selected, if  $pv_{[i]}$  does not exceed the significance level of 5%.

The soft thresholding methods use optimisation techniques to rank and select indicators. The LASSO method solves the following constrained minimisation problem

$$\min_{b} RSS + \lambda \sum_{i=1}^{M} |\xi_i|,$$

(4)

where *RSS* is the residual sum of squares from the regression of the variable of interest on all available regressors, i.e. the *N* candidate indicators and predetermined variables, with regression coefficients  $\xi_i$ . Parameter  $\lambda$  ( $\lambda \ge 0$ ) determines the degree of shrinkage, i.e. the extent to which coefficients are estimated as zero and therefore the corresponding indicators are not selected as relevant for the variable of interest. Thus, the LASSO regression starts from a large number of regressors and through the minimisation problem in (4), which takes into account cross-correlations among regressors, sets to zero the coefficients of regressors that are not informative for the variable of interest. The LARS method is a more general selection algorithm than LASSO, and is computationally fast (Efron et al., 2004; Bai & Ng, 2008). The LARS method is based on the idea of updating the projection of the variable of interest on a set of k - 1 predictors with information from a newly added predictor in the active

<sup>&</sup>lt;sup>5</sup> The deterministic terms include a constant and, if needed, dummy variables; the number of lags in the regression is determined by information criteria (Bayesian and Akaike).

set. The LARS method starts with all coefficients set to zero, picks the predictor with the highest correlation with the variable of interest and computes the projection; next, the algorithm searches for the predictor that is the most correlated with the current residual to update the projection. LARS then continues in an equiangular direction between the two selected variables until another predictor is added in the most correlated set and so on. After *k* steps there are *K* variables in the active set, i.e. there are *K* predictors with non-zero coefficients and the remaining variables have zero coefficients. The algorithm returns a set of indices that show the order according to which each variable joined the active set.

The three predictor selection methods described above are applied using recursive estimation samples, starting from the first sample of  $T_0$  observations, performing estimation and indicator selection before increasing the sample size by one observation (i.e. quarter) in the next iteration; thus, indicators are selected anew in each iteration.<sup>6</sup> Moreover, the three selection methods are applied separately for each variable of interest and over the different forecast horizons. The selection methods are used to rank the predictors in the dataset according to their information content with respect to GDP and each of the production-side components.

We consider two alternative schemes for isolating the informative predictors ordered through hard and soft thresholding: (a) selection in each estimation iteration and therefore every quarter (t), i.e. for each estimation sample, and (b) selection across all iterations, i.e. over all estimation samples.<sup>7</sup> In scheme (a), an indicator is selected, if it is ranked among the first 20 predictors with the lowest p-values under hard thresholding, or, is ordered among the best 20 predictors according to the LASSO or LARS algorithm; in this way, the size of the selected set is kept constant over sample iterations.<sup>8</sup> In scheme (b), an indicator is picked if the criteria described in scheme (a) hold, and its frequency of being selected, by each method, across all samples is ranked among the 20 highest. Thus, scheme (b) isolates the best 20 predictors that are selected most often across all iterations; the lowest selection frequency is around 30%.9 Both schemes result in a subset of 20 selected indicators for each thresholding method applied; in scheme (a) the selected predictors may differ in each estimation sample, while in scheme (b) the indicators chosen are the same for all estimation samples. Studies employing European data for forecasting growth find that the performance is favoured when the number of selected predictors is rather small (e.g. Caggiano, Kapetanios & Labhard 2011; Girardi, Golinelli & Pappalardo, 2017). The empirical analysis of Caggiano, Kapetanios and Labhard (2011) for large euro area countries and the UK shows that the highest forecasting gains are achieved when factors are estimated from quite small panels of preselected predictors, with sizes ranging from 12 (UK) to 22 (Italy) indicators.

The nature of the selected predictors across all sample iterations is outlined in Table 2 in the case of GDP. Table 2 shows the distribution of the 20 most frequently selected predictors over different types of variables. For the one-quarter horizon, the LASSO

 $<sup>^{6}</sup>T_{0}$  equals 24.

<sup>&</sup>lt;sup>7</sup> Scheme (a) follows Bai and Ng (2008) and scheme (b) resembles the selection strategy in Bulligan, Marcellino and Venditti (2015), and Girardi, Golinelli and Pappalardo (2017).

<sup>&</sup>lt;sup>8</sup> The maximum number of selected indicators by LASSO and LARS cannot exceed the sample size, if the sample size is smaller than the number of candidate predictors, as in the empirical application.

<sup>&</sup>lt;sup>9</sup> The lowest frequency for which an indicator is selected under scheme (b) varies somewhat across the variables of interest; the cut-off frequencies range from 17% for the GVA component in real estate activities to 48% for the GVA component in financial and insurance activities (see also Table 2 and Table A3).

and LARS methods pick more domestic rather than foreign and international predictors; the variables chosen by the hard thresholding method are evenly divided between the domestic and foreign groups. For longer horizons all methods select a higher proportion of foreign as opposed to domestic predictors. Survey indicators are included in the sets of selected predictors by all methods at high proportions (over 30%) for all horizons. For the very short horizon, real activity and labour market variables are frequently picked, particularly by soft thresholding methods; as the horizon increases all methods tend to select more financial indicators and, to a smaller extent, price variables, and fewer real activity and labour market predictors. Furthermore, the three selection methods show a high degree of commonality with respect to the chosen indicators, which tends to increase with the forecast horizon. For the one-quarter horizon, about 40% of predictors picked by hard thresholding are the same as those selected by the soft thresholding methods, while this percentage exceeds 65% for the longest horizon. The LASSO and LARS methods result in subsets that have over 70% of the predictors in common.

The selection of more domestic as opposed to foreign indicators for the one-quarter horizon, especially by soft thresholding methods, is found for about half of the production-side components (Table A3). For a number of components, foreign variables tend to dominate the set of selected predictors; these are components that relate to sectors of economic activity with dependency on external demand, for example transport and hospitality, professional services, and entertainment and recreation. For the majority of components, the sets of selected predictors consist mostly of financial indicators, followed by survey and real activity variables, especially for shorter horizons. Overall, there are no striking differences among the various selection methods applied; the three methods pick indicators from the different categories with similar proportions.

The last row of Table 2 gives the lowest frequency with which an indicator should be picked across sample iterations by a given method, in order to be included in the set of selected predictors under scheme (b). Across all sample iterations, the 20 best indicators are selected more frequently when the hard thresholding method is used; thus, for a given cut-off frequency the hard thresholding rule tends to select more indicators than LASSO and LARS. The cut-off frequencies do not considerably vary over the forecast horizon. Similar patterns are observed for the GDP components (Table A3). For some components, particularly those for which relevant predictors are not readily available (e.g. agriculture, real estate activities), the cut-off frequencies associated with LASSO and LARS are just below 20%, but for most of the components and horizons the cut-off frequencies are around or higher than 30%.

i creentages of selected pr	culeto	.10 e j	<i>ype</i> ,	und eu		reque	ileieo,	001	
Horizon (quarters)		1			4			8	
Predictor selection method	F-test	LASSO	LARS	F-test	LASSO	LARS	F-test	LASSO	LARS
Types of predictors									
Domestic	50	75	68	27	35	33	19	38	30
Foreign/international	50	25	32	73	65	67	81	62	70
By category									
Real activity and labour market	32	50	41	5	25	24	5	10	5
Financial	18	15	18	41	20	24	43	38	45
Prices	0	0	0	14	20	19	14	14	15
Business and consumer surveys	50	35	41	41	35	33	38	38	35
Cut-off frequency (%)	38	33	35	41	33	30	42	29	32

TABLE 2 Percentages of selected predictors by type, and cut-off frequencies. GDP

Notes: The percentages and frequencies are computed across all sample iterations and relate to scheme (b). The cut-off frequency shows the lowest frequency with which an indicator should be chosen across all sample iterations (i.e. all quarters) by a given method in order to be included in the set of selected predictors.

#### 4. Forecasting

#### 4.1 Forecasting exercise and overall performance

The panels of predictors selected by hard (F-test) and soft (LASSO and LARS) thresholding methods are used to extract factors and compute forecasts through the model described by equations (1) - (3). As the database includes a large number of monthly indicators, many of which are released at the end of the reference month, (e.g. financial and survey data), we can exploit information from leading indicators that falls within the forecast horizon. Thus, we extend the forecasting equation to incorporate the available forward-looking information in terms of leads, i.e.

$$\hat{y}_{T+h}^{h} = \hat{\alpha}_{h} + \sum_{j=0}^{p} \hat{\beta}_{h,j}' \hat{F}_{T-j} + \sum_{m=0}^{q} \hat{\gamma}_{h,m} y_{T-m} + \hat{\delta}_{h}' \hat{F}_{T+1}^{L},$$
(5)

where  $\hat{F}_{t+1}^{L}$  is an  $l \times 1$  vector of factors extracted from the panel of selected indicators with available data for some or all of the months in quarter t + 1.  $\hat{F}_{t+1}^{L}$  is estimated by principal components. In the empirical application,  $\hat{y}_{t+h}^{h}$  is the annualised growth rate defined as

$$\hat{y}_{t+h}^h = \frac{400}{h} \ln\left(\frac{Y_{t+h}}{Y_t}\right),$$

where  $Y_t$  denotes the level of GDP or the level of a production-side component.

We employ different specifications of the factor-augmented forecasting equation (5), depending on the choice of the number of factors, r, lags, p and q, and factor leads, l. The number of factors and the lag length of  $\hat{F}_t$  and  $y_t$  are chosen by the Akaike Information Criterion (AIC). An alternative specification of the forecasting equation is

obtained by applying the Bayesian Information Criterion (BIC) for determining the number of factors and lags in (5).<sup>10</sup> Following Caggiano, Kapetanios and Labhard (2011) the number of factors is determined by AIC or BIC, thus, taking into account the variable of interest; this is in contrast to the use of other information criteria that choose the number of factors that best summarises the panel of predictors, without reference to the variable to be forecast. Caggiano, Kapetanios and Labhard (2011) argue in favour of using information criteria to select the number of factors within the forecasting model, as opposed to criteria that rely solely on the variation of predictors in the panel, since not all factors chosen by the latter criteria are necessarily informative for forecasting the variable of interest. Moreover, in each of the AIC and BIC specifications the number of factor leads is set to l = 1 or l = 4, resulting in four different forecasting equations in total.

The pseudo out-of-sample forecasting exercise uses data over the period 1995Q1 – 2019Q4; the first estimation sample consists of the first 24 observations and the sample is extended by one quarter in each iteration. The factors and forecasting equations are estimated anew in each iteration, using the panels of predictors selected according to scheme (a) and scheme (b). Forecasts for the growth rate of GDP and the growth rates of the production-side components are computed for h = 1, ..., 8, using a pseudo out-of-sample set-up that mirrors the availability of monthly indicators in real time; the forecasts are evaluated over the period 2002Q3 – 2019Q4.

The forecasting equation (5) that includes factors, as well as autoregressive terms of  $y_t$ , is used to compute the factor-augmented autoregressive (FAR) forecasts. The FAR forecasts from the four specifications are combined using two alternative forecast combinations: (i) the mean, and (ii) the discounted mean squared forecast error (DMSFE), with a discounting factor set to 0.90 (see e.g. Stock & Watson, 2004, 2006). Furthermore, the resulting combination forecasts for the 11 components of GDP are aggregated to compute bottom-up GDP growth forecasts.<sup>11</sup>

For comparison purposes, we compute the mean and DMSFE combinations of forecasts from Autoregressive Distributed Lag (ADL) models, using as predictors, one at a time, indicators selected through schemes (a) and (b). More specifically, the ADL forecasting equation takes the form,

$$\hat{y}_{T+h}^{h} = \hat{a}_{h} + \sum_{j=0}^{p} \hat{b}_{h,j} X_{s,T-j} + \sum_{m=0}^{q} \hat{c}_{h,m} y_{T-m} + \hat{d}_{h} X_{s,T+1}^{L},$$
(6)

where  $X_{s,t}$  is a selected predictor according to scheme (a) or (b), and  $X_{s,t+1}^{L}$  denotes the lead, i.e. data on predictor  $X_{s,t}$  available within the forecast horizon. The lag lengths of  $X_{s,t}$  and  $y_t$  are chosen by AIC or BIC as in the factor-augmented equation (5).

Table 3 presents the root mean squared forecast error (RMSFE) of combinations of FAR and ADL model forecasts for GDP growth relative to that of the benchmark AR(1) model. The table also shows the relative RMSFE for GDP growth bottom-up forecasts, obtained by aggregating combinations of FAR and ADL component forecasts. The results are given for the three selection methods employed, i.e. hard thresholding (F-

<sup>&</sup>lt;sup>10</sup> AIC or BIC can choose up to four factors; the upper bound for the number of factors was guided by information criteria that summarise the dataset (e.g. Bai & Ng, 2002) and sample size limitations. Furthermore, AIC or BIC can select up to two and four lags for  $\hat{F}_t$  and  $y_t$ , respectively. The results for the case where AIC or BIC can select up to four lags for both  $\hat{F}_t$  and  $y_t$ , when the sample size allows it, show some deterioration in the forecasting performance.

<sup>&</sup>lt;sup>11</sup> Bottom-up GDP growth forecasts are computed by aggregating the growth contributions of the components (for more details, see Eurostat, 2013, Ch. 6).

test), LASSO and LARS under schemes (a) and (b). For comparison purposes, the column labelled "All predictors" presents the relative RMSFE when no selection method is applied, i.e. all predictors in the dataset are used to extract the factors included in the FAR models, and all indicators, one at a time, are included in the ADL models.

					,							
Horizon (quarters)		-	1				4			;	8	
Predictor selection method	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors
Benchmark AR(1): RMSFE		1.	01			3.	91			4.	94	
Scheme (a): each t												
FAR - mean	0.83	0.89	0.98	0.91	0.76	0.69	0.69	0.80	1.01	1.04	1.02	0.99
FAR - DMSFE	0.82	0.89	0.99	0.90	0.76	0.68	0.69	0.80	1.00	1.04	1.02	0.99
FAR - mean, bottom-up	1.06	0.92	1.05	0.86	0.81	0.74	0.75	0.86	1.19	1.29	1.30	1.40
FAR - DMSFE, bottom-up	1.06	0.91	1.09	0.85	0.80	0.74	0.75	0.86	1.19	1.29	1.30	1.39
ADL - mean	0.83	0.81	0.82	0.87	0.88	0.82	0.84	0.93	1.00	0.97	0.96	1.02
ADL - DMSFE	0.76	0.72	0.75	0.80	0.87	0.82	0.86	0.93	1.02	1.08	0.98	1.01
ADL - mean, bottom-up	0.90	0.91	0.91	0.94	1.10	0.89	0.90	0.95	1.08	1.11	1.11	1.11
ADL - DMSFE, bottom-up	0.91	0.90	0.85	0.92	1.11	0.89	0.88	0.95	1.01	1.07	1.12	1.11
Scheme (b): all t												
FAR - mean	0.82	0.71	0.75	0.91	0.56	0.56	0.49	0.80	0.85	0.92	0.90	0.99
FAR - DMSFE	0.82	0.71	0.75	0.90	0.55	0.56	0.49	0.80	0.85	0.92	0.90	0.99
FAR - mean, bottom-up	0.80	0.81	0.77	0.86	0.74	0.53	0.51	0.86	1.28	1.04	0.95	1.40
FAR - DMSFE, bottom-up	0.80	0.80	0.75	0.85	0.74	0.53	0.51	0.86	1.29	1.04	0.93	1.39
ADL - mean	0.77	0.80	0.80	0.87	0.75	0.81	0.80	0.93	0.86	0.90	0.90	1.02
ADL - DMSFE	0.72	0.72	0.72	0.80	0.77	0.80	0.79	0.93	0.84	0.89	0.88	1.01
ADL - mean, bottom-up	0.86	0.88	0.87	0.94	0.83	0.85	0.84	0.95	0.95	1.02	0.98	1.11
ADL - DMSFE, bottom-up	0.83	0.84	0.82	0.92	0.81	0.82	0.82	0.95	0.92	1.01	0.96	1.11

TABLE 3 Relative RMSFE, GDP

Notes: The table entries show the RMSFE of each forecast method relative to that of the AR(1) model; the pseudo out-of-sample forecasts are computed using subsets of predictors selected by hard (F-test) or soft (LASSO, LARS) thresholding methods, or the full set of predictors in the dataset. Entries in bold indicate statistical significance at 10% level for the modified Diebold-Mariano test of equal forecast accuracy (Diebold & Mariano, 1995; Harvey, Leybourne & Newbold, 1997). Under scheme (a), predictors are selected for each estimation sample (i.e. in each estimation and forecast iteration), while under scheme (b) predictors are chosen over all estimation samples and are therefore the same in all estimation and forecast iterations.

Keeping the set of optimal predictors constant for each estimation sample as in scheme (b) results in improved forecasting performance vis-à-vis selecting the best predictors for each estimation sample as in scheme (a). Preselection, especially when it results in constant subsets of predictors over estimation samples (scheme (b)), leads to higher forecast accuracy, compared to the case in which no preselection of indicators is applied, i.e. the full dataset is employed for factor estimation and forecasting. Forecast gains compared to the benchmark reach 30% and 50% under scheme (a) and scheme (b), respectively. Under scheme (a), statistically significant gains are mainly

concentrated in the middle of the horizon, while for one-quarter ahead forecasts the benefits of preselection are mostly associated with ADL combination forecasts. Under scheme (b), combinations of FAR or ADL models for GDP with predictors preselected either through hard or soft thresholding methods outperform both the benchmark and combinations based on the full panel of predictors, for all horizons considered.

Bottom-up GDP growth forecasts based on component forecasts computed using predictors specifically selected for each component under scheme (b), attain statistically significant gains for horizons of one and four quarters. Furthermore, for short horizons, bottom-up growth forecasts obtained from component FAR models with preselected predictors outperform bottom-up forecasts based on either all predictors or preselected predictors employed in component ADL models. In sum, the forecasting performance depends on (i) the horizon, with the largest gains achieved for h = 4, and, most importantly, (ii) whether the subsets of chosen predictors remain constant or change over the estimation samples. The technique employed for preselecting the predictors and the forecast combination applied (mean vs. DMSFE) are not found to considerably affect the forecasting performance.

Examining the forecasting performance at the GDP component level, we find that not all components are associated with significant gains over the simple AR benchmark (Table A4). Predictors do not seem to contain useful information for forecasting activity developments in the sectors of agriculture, information and communication, and real estate activities. For the remaining components, employing economic and financial indicators tends to improve the forecasting performance over the benchmark. More specifically, for the majority of components, gains are achieved with predictor preselection under scheme (b), i.e. when the sets of preselected predictors remain unchanged over estimation and forecasting iterations, a result also found for GDP. In the sectors of construction, as well as trade, transportation, accommodation and food services, GVA growth forecasts obtained from sets of preselected indicators outperform forecasts based on the full panel of predictors; preselection reduces the RMSFEs for all horizons, particularly in the case of the LASSO and LARS selection rules. Preselection under scheme (b) results in statistically significant gains over the benchmark for horizons up to four quarters in the industry sector; the gains, however, are close to those registered without preselection. For the rest of the services components (i.e. financial, professional, public, education, health and other services), some gains from preselection vis-à-vis the benchmark are found under scheme (b), but are not uniformly significant across selection methods, forecasting models and horizons; nevertheless, models with preselected predictors tend to result in lower RMSFE than those using the full panel. Finally, using preselected predictors for forecasting the net taxes component significantly improves on the benchmark and lowers the RMSFE compared to employing the full panel of predictors for forecasting. The highest gains from preselection over the benchmark (up to 51%) are attained for the construction, trade and net taxes components, while smaller gains (up to 17%) are found for the industry sector, as well as for financial and professional services, public administration, education, health and, other services. Forecasting with constant sets of preselected predictors improves the forecast accuracy for the majority of GDP components; the improvement is then reflected in the precision of the bottom-up GDP growth forecasts, especially for horizons up to four quarters.

#### 4.2 Forecasting performance: crisis vs. normal times

This section examines whether the forecasting performance in the case of GDP growth varies depending on the state of the economy. As discussed in the previous section, using sets of preselected indicators in FAR and ADL models outperforms the AR benchmark significantly and results in higher forecasting gains vis-à-vis employing the full dataset of predictors. In this section, we investigate whether the RMSFEs obtained from the different predictor selection techniques and forecasting methods change substantially between crisis and normal times, and whether higher accuracy compared to the benchmark is maintained over the business cycle. Since the forecasting evaluation period is small, the analysis is descriptive in nature.

The performance of the methods considered for forecasting GDP growth is evaluated over two separate periods: (i) a crisis period, which includes the global financial crisis, the Greek debt crisis, and the ensuing crisis in Cyprus that peaked in 2013, and (ii) the rest of the sample, namely a period of steady growth (no crisis).<sup>12, 13</sup>

Figure 1 shows the RMSFEs for forecast combinations of FAR and ADL model forecasts relative to the AR benchmark. The figures report the relative RMSFEs over the full sample and the two sub-samples covering the crisis and no crisis periods; the RMSFEs are obtained using all predictors in the dataset for forecasting, as well as sets of indicators preselected through hard (F-test) and soft thresholding (LASSO and LARS) rules. We report the results for DMSFE combinations as this method penalises forecasts through past errors. Based on the findings concerning the forecasting performance over the full period (section 4.1), we focus on scheme (b), in which the sets of selected predictors remain constant over the estimation samples; scheme (b) results in considerably higher overall accuracy compared to scheme (a), namely the case of preselecting predictors in each estimation sample. The lower part of Figure 1 depicts the RMSFEs for bottom-up GDP growth forecasts obtained from aggregating component growth forecasts.

Looking at the crisis subsample, combinations of FAR or ADL model forecasts from preselected predictors outperform both the AR benchmark and the corresponding combinations that rely on the full panel of predictors; this result also holds for the full forecasting period. The largest gains from preselection during crisis reach 50% and 60% vis-à-vis the benchmark for horizons of one and four quarters, respectively, but decline to just above 10% for the eight-quarter horizon. The use of preselected predictors in FAR models to directly forecast GDP growth appears to be an optimal strategy for short-term forecasting ( $h \le 4$ ) during the crisis period; bottom-up growth forecasts based on FAR models and preselected predictors for GDP components also result in reliable short-term forecasts in the crisis period. The forecast accuracy for the longest horizon can be benefitted from predictor preselection during turbulent periods, but to a limited extent. During normal times (no crisis), predictor preselection leads to larger improvements in forecast precision for longer horizons, especially for h = 8, than for the one-quarter horizon. Applying indicator selection with respect to

<sup>&</sup>lt;sup>12</sup> The Greek debt crisis affected Cyprus mainly because of links between the banking systems in the two countries at the time.

<sup>&</sup>lt;sup>13</sup> The crisis period includes the following quarters 2008Q3 – 2009Q4 and 2011Q3 – 2015Q1; these are quarters in which one of the following holds: (i) the quarter-on-quarter growth rate is negative, following a quarter with a negative quarter-on-quarter growth rate; (ii) the quarter-on-quarter growth rate is negative quarter-on-quarter growth rate; be a quarter with a positive quarter-on-quarter growth rate is positive, following a quarter with a positive quarter-on-quarter growth rate is positive and the change in the unemployment rate is positive, following a quarter with a negative quarter-on-quarter growth rate is positive and the change in the unemployment rate is positive quarter-on-quarter growth rate.

production-side components and computing bottom-up forecasts for GDP growth, results in the best performing forecasting strategy for the four- and eight-quarter horizons in normal times. Moreover, preselection improves forecasting precision for the shortest horizon in normal times, when the indicators are used in ADL models for GDP growth. Gains from preselection, vis-à-vis employing the full panel of predictors for forecasting, are larger in the crisis subsample than during normal times, averaging about 18% across methods and horizons; still, preselection leads to RMSFE reduction of about 14% on average in the no crisis subsample. Overall, preselection during the crisis period generates large forecast gains for horizons up to four quarters, while preselection in normal times benefits forecast accuracy for longer horizons.



FIGURE 1

### 5. Concluding remarks

The use of large datasets for macroeconomic forecasting has spurred research on the effects of preselecting predictors on the forecasting performance. A number of empirical studies document gains when a subset of predictors, as opposed to the full dataset, is employed for forecasting. This paper uses a dataset of 200 indicators and applies a hard thresholding method, based on the F-test, as well as the LASSO and LARS algorithms in order to select indicators from the dataset. The selected indicators are subsequently used for forecasting economic activity in Cyprus at the aggregate and sectoral levels.

The subsets of predictors are selected separately for each variable of interest, namely GDP and the 11 production-side components; also, the sets of selected predictors are allowed to vary over the forecast horizon. Predictor selection is carried out through two alternative schemes: (a) in each quarter, i.e. for each estimation sample, and (b) across all quarters, i.e. over all estimation samples. For the one-quarter horizon, the sets of selected predictors tend to contain a higher proportion of domestic as opposed to foreign predictors, while the opposite occurs for longer horizons. In the case of GDP, indicators from the group of business and consumer surveys are picked in high proportions for all horizons and by all thresholding techniques. For most of the production-side components, the sets of selected predictors contain mainly financial indicators, and, to a smaller degree, survey and real activity variables.

The performance of GDP and component growth forecasts computed from the selected sets of predictors is juxtaposed to that of forecasts constructed using the full dataset. Preselection leads to higher forecast accuracy compared to the case in which the full dataset is employed for forecasting, particularly when the sets of selected predictors remain constant over time. The thresholding technique employed for preselecting predictors is not found to substantially affect the forecasting performance. The benefits from preselecting predictors prior to forecasting GDP growth are enhanced during a crisis period, particularly for horizons up to four quarters. During normal times, forecast gains from preselection can also be attained vis-à-vis employing the full panel of predictors, especially when the forecasts are computed for longer horizons.

The results of the paper highlight the importance of timely available predictors, particularly survey and financial indicators, for forecasting economic activity in Cyprus. Narrowing down a large database to a smaller set of timely published indicators delivers a computationally fast tool for constructing nowcasts and monthly updates of forecasts. This tool can be particularly useful during times of turbulence in the economy.

### Acknowledgements

The Economics Research Centre of the University of Cyprus acknowledges funding from the European Union, the Ministry of Finance and the University of Cyprus for conducting the project "Business and Consumer Surveys" in Cyprus.

# Appendix

# TABLE A1 List of dependent variables

Variable description	NACE Revision 2 classification	Abbreviation
Gross Domestic Product		GDP
Gross Value Added		GVA
1. Agriculture, forestry and fishing	А	AGR
<ol><li>Mining and quarrying; manufacturing; electricity, gas, steam and air conditioning supply; water supply; sewerage, waste management and remediation activities</li></ol>	B, C, D, E	IND
3. Construction	F	CON
4. Wholesale and retail trade; repair of motor vehicles and motorcycles; transportation and storage; accommodation and food service activities	G, H, I	TRA
5. Information and communication	J	INF
6. Financial and insurance activities	K	FIN
7. Real estate activities	L	REA
8. Professional, scientific and technical activities; administrative and support service activities	M, N	PRO
9. Public administration and defence; compulsory social security; education; human health and social work activities	O, P, Q	PEH
10. Arts, entertainment and recreation, other services	R, S, T, U	OTH
Taxes less subsidies on products		TAX

Note: The variables refer to seasonally adjusted, chain-linked volume measures; the data were obtained from the Statistical Service of Cyprus.

# TABLE A2

# List of predictors in the dataset

Weights description		Details	
variable description	А	В	С
Registration of motor vehicles (number of passenger cars)	D	М	3
Registration of motor vehicles (number of all vehicles)	D	Μ	3
Volume index of retail trade - NACE 47, (2015=100)	D	Μ	3
Volume index of retail trade - NACE 47 excl. 47.3, (2015=100)	D	Μ	3
Building permits authorised (actual number)	D	Μ	3
Building permits authorised (value), deflated	D	Μ	3
Volume index of manufacturing production, (2015=100)	D	Μ	3
Index of industrial production, (2015=100)	D	Μ	3
Total imports/arrivals (c.i.f.), deflated	D	Μ	3
Total exports/dispatches (f.o.b.), deflated	D	Μ	3
Tourist arrivals in Cyprus	D	Μ	3
Tourist arrivals in Cyprus from the United Kingdom (usual residency)	D	Μ	3
Tourist arrivals in Cyprus from Germany (usual residency)	D	Μ	3
Tourist arrivals in Cyprus from Russia (usual residency)	D	Μ	3
Tourist arrivals in Cyprus from Greece (usual residency)	D	Μ	3
Tourism revenues in Cyprus (act €mn), deflated	D	Μ	3
Local cards used in Cyprus (sales value), deflated	D	Μ	3
Foreign cards used in Cyprus (sales value), deflated	D	Μ	3
Total number of property sale contracts (registered contracts, Department of Lands and Surveys)	D	Μ	3
Total number of property transfers (number of parcels, Department of Lands and Surveys)	D	Μ	3
Total value of property transfers (accepted amount, Department of Lands and Surveys), deflated	D	Μ	3
Registration of new companies in Cyprus (number)	D	Μ	3
VAT on products, deflated	D	Μ	3
Total registered unemployed (actual number)	D	Μ	3
Registered unemployed - Agriculture, forestry and fishing (actual number)	D	Μ	3
Registered unemployed - Manufacturing (actual number)	D	Μ	3
Registered unemployed - Construction (actual number)	D	Μ	3
Registered unemployed - Wholesale & retail trade (actual number)	D	Μ	3
Registered unemployed - Restaurants & hotels (actual number)	D	Μ	3
Registered unemployed - Transport, storage & communication (actual number)	D	Μ	3
Registered unemployed - Finance, insurance, real estates & business services (actual number)	D	Μ	3
Registered unemployed - Newcomers (actual number)	D	Μ	3
Unemployment rate	D	М	2

# TABLE A2 (continued)

Valuation uses of provide section of the se	Variable description		Details	
Cyrpus Sack Exchange, All Share General Composite Index         D         M         1           Cyrpus Sack Exchange, ITSBL (SE 20 Index         D         M         3           Cyrpus Sack Exchange, ITSBL (SE 20 Index         D         M         3           Cyrpus Sack Exchange, ITSBL Index         D         M         3           Cancers Trats on Agencia for the Index         D         M         2           Interest rates on Agencia for the Index Index         D         M         2           Interest rates on Agencia for the Index Index         D         M         2           Interest rates on Agencia for households (new basiness), overnight         D         M         2           Interest rates on Agencia for households (new basiness), overnight         D         M         2           Interest rates on Agencia for households (new basiness), overnight         D         M         2           Interest rates on Assor of non-financial corporations (new basiness), overnight rate up to 1 year initial fination         D         M         2           Interest rates on Assor of non-financial corporations (new basiness), overnight rate up to 1 year initial fination         D         M         2           Interest rates on Assor of households (new basines), overnight rates         D         M         1           Interest ra	variable description	А	В	С
Cyrum Sock Exchange, All Share General Composite Index, historical volatility         D         M         3           Cyrum Sock Exchange, Harde Index         D         M         3           Cyrum Sock Exchange, Harde Index         D         M         3           Externant exter on deposite of non-financial corporations (new busines), naturity up to 1 year         D         M         2           Interest rates on deposite of non-financial corporations (new busines), naturity up to 1 year         D         M         2           Interest rates on deposite of non-financial corporations (new busines), maturity up to 1 year         D         M         2           Interest rates on deposite of non-financial corporations (new busines), or 200 months         D         M         2           Interest rates on loans of non-financial corporations (new busines), or 200 FIR nut - floating rate up to 1 year initial fination         D         M         2           Interest rates on loans of non-financial corporations (new busines), or 200 FIR nut - floating rate up to 1 year initial fination         D         M         2           Interest rates on loans of non-financial corporations (new busines), or 200 FIR nut - floating rate up to 1 year initial fination         D         M         2           Interest rates on loans of non-financial corporations (new busines), non-trinking rate up to 1 year initial fination         D         M         1	Cyprus Stock Exchange, All Share General Composite Index	D	М	3
Cypren Stack Exchange, Hole Index Cypren Stack Exchange, Hole Index Deposite of non-infancial corporations (new bisines), maturity up to 1 year Interest rates on deposite of households (new busines), overright Interest rates on deposite of households (new busines), overright Interest rates on deposite of households (new busines), redex rule at noice up to 3 months Interest rates on deposite of households (new busines), and the year D M 2 Interest rates on leaves of households (new busines), Park Overfants Interest rates on leaves of households (new busines), Park Overfants Interest rates on leaves of households (new busines), Park Overfants Interest rates on leaves of households (new busines), Park Dyname, Rothing rate up to 1 year initial fixation D M 2 Interest rates on leaves of households (new busines), Park Dyname, Rothing rate up to 1 year initial fixation D M 2 Interest rates on leaves of households (new busines), House Internet, Rothing rate up to 1 year initial fixation D M 2 Loaves to non-MTE (outstanding amounts), Intarting up to 1 year initial fixation D M 1 Interest rates on leaves of households (new busines), Households, park new leaves (fam) D M 3 Volumes of new leave leave and non-financial corporations fee user area households, park new leaves (fam) D M 3 Volumes of new leave leave and non-financial corporations, fee user park househoused, park new leaves (fam) D M 3 Volumes of new leave leave and non-financial corporations, fee user area households, park new leaves (fam) D M 3 Firm chales of course new rane non-financial corporations, fee user area households, park new leaves (fam) D M 3 Firm chales of course new rane non-financial corporations, fee user area households, park new leaves (fam) D M 3 Firm chales of course new rane households, park new leaves (fam) D M 3 Firm chales for leaves ar	Cyprus Stock Exchange, All Share General Composite Index, historical volatility	D	Μ	1
<ul> <li>Cyprus Bock Exchange, Investment Companies Index.</li> <li>Cyprus Bock Exchange, Investment Companies Index.</li> <li>Cyprus Bock Exchange, Investment Companies Index.</li> <li>Different attes on deposits of non-financial corporations (new basines), maturity up 1 year</li> <li>Different attes on deposits of non-financial corporations.</li> <li>Interest rates on bases of non-financial corporations.</li> <li>Interest rates on ba</li></ul>	Cyprus Stock Exchange, FTSE/CySE 20 Index	D	М	3
C prite shock is including and the solution of the solutions) overnight [10] and [10	Cyprus Stock Exchange, Hotels Index	D	M	3
Initial intervention of a possibility of an analysis providing intervention of a possibility of a poss	Cyprus Stock Exchange, Investment Companies Index	D	M	3
Interest rates on deposits of households (new business), noternable at notice up to 3 months D M 2 2 Interest rates on deposits of households (new business), indernable at notice over 3 months D M 2 2 Interest rates on deposits of households (new business), indernable at notice over 3 months D M 2 2 Interest rates on loans of non-financial corporations (new business), busine vertification on loans of non-financial corporations (new business), busine vertification on loans of non-financial corporations (new business), busine vertification on loans of non-financial corporations (new business), busine vertification on loans of non-financial corporations (new business), construct redit , not Rull rate in the non-financial corporations (new business). Finaling rate up to 1 year initial fination D M 2 2 Instruct rates on loans of households (new business), including rate up to 1 year initial fination D M 2 2 Loans to non-Mil's (outstanding amounts), intervity yea 1 year initial fination D M 2 2 Loans to non-Mil's (outstanding amounts), intell, year to the loans of households (prev hismess), lending for house purches, finating rate up to 1 year initial fination D M 2 2 Loans to non-Mil's (outstanding amounts), intal, yrup to 1 year initial fination D M 2 2 Loans to non-Mil's (outstanding amounts), intal, yrup to 1 year initial fination D M 2 2 Loans to non-Mil's (outstanding amounts), intal, yrup to 1 year initial fination D M 2 2 Loans to non-Mil's load tarking the (substanding amounts), intal, yrup to 1 year initial fination D M 2 2 Loans to non-Mil's load tarking the (substanding amounts), intal, yrup to 1 year initial fination D M 2 2 Loans to non-Mil's load tarking new loans (ferm) D M 4 3 Volumes of new loans to true on an on-financial corporations, purchess (ferm) D M 4 3 Volumes of new loans to true on an on-financial corporations (ferm) D M 4 3 1 Loan-to-deposit ratio: total loans to total drappendic previous (ferm) D M 4 3 1 Volumes of new loans to true on an on-financial corporations, purchess (ferm) D M 4	Interest rates on deposits of non-financial corporations (new business), overnight	D	M	2
Interest rates on deposits of households (note husiness), nodemplie a totice up to 3 monthsDM2Interest rates on deposits of households (note husiness), maturity up to 1 yearDM2Interest rates on deposits of households (new business), and varity up to 1 yearDM2Interest rates on loans of on-financial corporations (new business), and varity up to 1 year initial finationDM2Interest rates on loans of households (new business), consumer credit, floating rate up to 1 year initial finationDM2Interest rates on loans of households (new business), onther CHUS 1 nmfloating rate up to 1 year initial finationDM2Interest rates on loans of households (new business), onther (nehng, floating rate up to 1 year initial finationDM2Interest rates on loans of households (new business), onther (nehng, floating rate up to 1 year initial finationDM2Loans to non-Miles hold (neb husiness), onther (nehng, floating rate up to 1 year initial finationDM2Loans to non-Miles hold (neb husiness), onther (nehng, floating rate up to 1 year initial finationDM1Loans to non-Miles hold (neb husiness), onther (nehng, floating rate up to 1 year initial finationDM1Loans to non-Miles hold (neb husiness), and any to 1 yearDM11Loans to non-Miles hold (neb husiness), and any to 1 yearDM11Loans-deeposits of non-Miles hold (neb husiness), and any to 1 yearDM3Voltames of new loans to toul adpositsD </td <td>Interest rates on deposits of households (new business), overnight</td> <td>D</td> <td>M</td> <td>2</td>	Interest rates on deposits of households (new business), overnight	D	M	2
Interest rates on deposits of households (ore standing amounts), redeemable at notice over 3 months Interest rates on deposits of households (ore business), naturity up to 1 year Interest rates on loans of non-financial corporations (new business), our URI 1 mm - Hoating rate up to 1 year initial fixation Interest rates on loans of non-financial corporations (new business), our EUK 1 mm - Hoating rate up to 1 year initial fixation Interest rates on loans of non-financial corporations (new business), our EUK 1 mm - Hoating rate up to 1 year initial fixation Interest rates on loans of households (new husiness), leading for house purches, floating rate up to 1 year initial fixation Interest rates on loans of households (new husiness), leading for house purches, floating rate up to 1 year initial fixation Interest rates on loans of households (prew husiness), leading for house purches, floating rate up to 1 year initial fixation Interest rates on loans of households (prew husiness), leading for house purches, floating rate up to 1 year initial fixation Interest rates on loans of households (prew husiness), leading for house purches, floating rate up to 1 year initial fixation Interest rates on loans of households (prew husiness), purce resolutes, floating rate up to 1 year initial fixation Interest rates on loans of households (prew husiness), purce resolutes, floating rate, purcest on non-MFIs hold with MFIs (outstanding amounts), loating rule of the set (Run) Interest rates on loans to total adopsets Interest rates on loans of thouse husines on non-financial corporations, purce new loans (Run) Interest rates on loans of households (prev new loans (Run) Interest rates on loans of households (Prev husine purchase, (Run) Interest rates on loans of households (Prev new loans (Run) Interest rates on loans of households (Prev new loans (Run) Interest rates on loans of households (Prev husine purchase, (Run) Interest rates on loans of households (Prev new loans (Run) Interest rates on loans of households (Prev new loans	Interest rates on deposits of households (new business), redeemable at notice up to 3 months	D	Μ	2
Interest rates on deposits of households (new busines), maturity up to 1 year         D         M         2           Interest rates on loans of non-financial corporations (new busines), over RUR 1 nm - floating rate up to 1 year initial floation         D         M         2           Interest rates on loans of non-financial corporations (new busines), over RUR 1 nm - floating rate up to 1 year initial floation         D         M         2           Interest rates on loans of households (new busines), loaning frate up to 1 year initial floation         D         M         2           Interest rates on loans of households (new busines), loaning frate up to 1 year initial floation         D         M         2           Learus to non-AFIE houtstholds (new business), annual growth rates         D         M         1           Learus to non-AFIE houtstholds (new business) annual growth rates         D         M         1           Deposits of non-AFIE houtstholds manual growth rates         D         M         1           Loans to hous-NEI houtstholds manual floating rate up to 1 year initial floating rates up to 1 year	Interest rates on deposits of households (outstanding amounts), redeemable at notice over 3 months	D	М	2
Interest rates on loans of non-financial corporations (new busines), put b UR 1mn - floating rate up to 1 year initial faution         D         M         2           Interest rates on loans of non-financial corporations (new busines) - over TUR 1mn - floating rate up to 1 year initial faution         D         M         2           Interest rates on loans of households (new business), consumer credit, Busting rate up to 1 year initial fixation         D         M         2           Interest rates on loans of households (new business), onburn (rating 10 year initial fixation         D         M         2           Interest rates on loans of households (new business), onburn (rating 10 year initial fixation         D         M         2           Interest rates on loans of households (new business), onburn (rate year loans in the loans households (new business), onburn (resolution)         D         M         1           Depositio for no-ME he hold with MES (outstanding amounts), onestic residents, annual growth rates         D         M         1           Depositio for no-ME he hold with MES (outstanding amounts), conset (resolution)         D         M         3           Volumes of new loans to ura anon-financial corporations & euro area households, pure new loans (from)         M         3           Volumes of new loans to ura ano households) pure new loans (from)         D         M         3           Volumes of new loans to ura ano households) pure new loans (from)	Interest rates on deposits of households (new business), maturity up to 1 year	D	Μ	2
Interest ates on loans of non-financial corporations (new business) - up to UVR 1 mm - floating rate up to 1 year initial fixation M 2 Interest rates on loans of non-shoulds (new business), loans purchase, floating rate up to 1 year initial fixation M 2 Interest rates on loans of nonsolubolds (new business), loans for house purchase, floating rate up to 1 year initial fixation M 2 Interest rates on loans of nonsolubolds (new business), loans for house purchase, floating rate up to 1 year initial fixation M 2 Interest rates on loans of nonsolubolds (new business), loans (rate up to 1 year initial fixation M 2 Interest rates on loans of nonsolubolds (new business), loans (rate up to 1 year initial fixation M 2 Loans to non-MFI hold with MFIs (outstanding amounts), loansetic residents, annual growth rates D M 1 Depositis of non-MFIs hold with MFIs (outstanding amounts), donsetic residents, annual growth rates D M 1 Loans to upon bours on an on-financial corporations & euro area households, pure new loans (fcm) D M 3 Volumes of new loans to our an non-financial corporations pure new loans (fcm) D M 3 Volumes of new loans to uero area households, pure new loans (fcm) D M 3 Volumes of new loans to uero area households, pure new loans (fcm) D M 3 Consumer Prices (HCP), 2015-100) D M 3 Price Index, 2015-100) D M 3 Consumer Prices (HCP), 2015-100) D M 3 Economic Sentiment Indicator (Economics Research Centre) M 2 Employment Expectations Indicator (Economics Research Centre) M 2 Erroices, Assessments of business situation orer the past 3 months, balance D M 2 Erroices, Assessments of business situation orer the past 3 months, balance D M 2 Erroices, Assessments of business situation orer the past 3 months, balance D M 2 Erroices, Stalling price expectations over the next 3 months, balance D M 2 Erroices, Stalling price expectations over the next 3 months, balance D M 2 Erroices, Stalling price expectations over the next 3 months, balance D M 2 Erroices, Stalling price expectations over the neat 3 months, balance D M 2 Er	Interest rates on loans of non-financial corporations (new business), bank overdrafts	D	Μ	2
Interest rates on hours of non-influcta icorporations (new tousiness) - over EUX 1 mm - fording rate up to 1 year initial fraction D M 2 Interest rates on hours of households (new business), consumer croßin, Botting rate up to 1 year initial fraction D M 2 Interest rates on hours of households (new business), consumer croßin, Botting rate up to 1 year initial fraction D M 2 Interest rates on hours of households (new business), consumer croßin, Botting rate up to 1 year initial fraction D M 2 Loans to non-MTe hold with MFs (outstanding amounts), outal, annual growth rates D M 1 Deposits of non-MTe hold with MFs (outstanding amounts), outal, annual growth rates D M 1 Loans to non-MTe hold with MFs (outstanding amounts), donestic residents, annual growth rates D M 1 Loans to non-MTe hold with MFs (outstanding amounts), donestic residents, annual growth rates D M 1 Loans to non-MTe hold with MFs (outstanding amounts), donestic residents, annual growth rates D M 1 Loans to non-MTe hold with MFs (outstanding amounts), donestic residents, annual growth rates D M 3 Volumes of new loans to euro area non-financial corporations & euro area households, pure new loans (fem) D M 3 Volumes of new loans to euro area non-financial corporations (euron area households, pure new loans (fem) D M 3 Consumer Price index, (2015-100) D M 3 Price Index of consumer Prices (HCP), (2015-100) D M 3 Price Index of consumer Prices (HCP), (2015-100) D M 3 Eronomic Sentiment Indicatorie Research Centre) D M 2 Eronomic Sentiment Indicatorie Research Centre) D M 2 Erorices, Assessments of business situation over the past 3 months, balance D M 2 Ervices, Sentiment Indicatorie Research Centre) D M 2 Ervices, Sentiment Indicatorie Control, balance D M 2 Ervices, Sentiment Indicatorie (term past 3 months, balance D M 2 Ervices, Sentiment on Advances and anoths, balance D M 2 Ervices, Entothese Rothinest of the next 3 months, balance D M 2 Ervices, Entothese Rothinest of the next 3 months, balance D M 2 Ervices, Floridhere Indicator (NACE E4, 65, 66)	Interest rates on loans of non-financial corporations (new business) - up to EUR1 mn - floating rate up to 1 year initial fixation	D	M	2
Interest rates of notes of nonsentials (see vubuies), leading for high pure part of section rates of nonsential fraction in the pure section of the section of nonsential (see vubuies), leading for high pure vubuies (noting rate up to 1) year initial fraction D M 2 interest rates on homes of households (see vubuies), montage and the pure vubuies (to 1) year initial fraction D M 1 2 Loars to non-MEs (sottstanding amounts), both annual growth rates D M 1 1 Deposits of non-MEs held with MEi (sottstanding amounts), donestic residents, annual growth rates D M 1 1 Deposits of non-MEs held with MEis (sottstanding amounts), donestic residents, annual growth rates D M 1 1 Deposits of non-MEs held with MEis (sottstanding amounts), donestic residents, annual growth rates D M 1 1 Deposits of non-MEs held with MEis (sottstanding amounts), donestic residents, annual growth rates D M 1 1 Deposits of non-MEs held with MEis (sottstanding amounts), donestic residents, annual growth rates D M 1 1 Deposits of non-MEs held with MEis (sottstanding amounts), donestic residents, annual growth rates D M 3 Volumes of new loans to curo area non-financial corporations, pure new loans (fcm) D M 3 Volumes of new loans to curo area non-financial corporations, pure new loans (fcm) D M 3 Volumes of new loans to curo area households, pure new loans (fcm) D M 3 Consume Price (lacks, 2015-100) D M 3 Consume Price	Interest rates on loans of hourshelds (country in the business) - over EUK 1 mm - floating rate up to 1 year initial fixation Interest rates on loans of hourshelds (country in sec), country and it floating rate up to 1 year initial fixation		M	2
Interest rates on hours of households (outsmining amount), mutainly up to 1 year initial fraktion D M 2 Loans to non-MEs (outstanding amounts), total, annual growth rates D M 1 Deposits of non-MEs held with MEs (outstanding amounts), total, annual growth rates D M 1 Deposits of non-MEs held with MEs (outstanding amounts), total, annual growth rates D M 1 Loans to non-MEs held with MEs (outstanding amounts), total, annual growth rates D M 1 Loans-to deposit ratio: total loans to total deposits residents, annual growth rates D M 1 Loan-to-deposit ratio: total loans to total deposits of non-MEs held with MEs (outstanding amounts), donestic residents, pure new loans (fem) D M 3 Volumes of new loans to curo area non-financial corporations & euro area households, pure new loans (fem) D M 3 Volumes of new loans to curo area households, pure new loans (fem) D M 3 Volumes of new loans to curo area households, pure new loans (fem) D M 3 Consumer Price Index, 2015-100) D M 3 Economic Sentiment Indicator (Economics Research Centre) D M 2 Exprojement Expectations Indicator (Economics Research Centre) D M 2 Errolpoyment Expectations Indicator (Economics Research Centre) D M 2 Errolex, Evolution of demand over the past 3 months, balance D M 2 Errolex, Evolution of demand over the past 3 months, balance D M 2 Errolex, Evolution of demand over the past 3 months, balance D M 2 Errolex, Evolution of demand over the next 3 months, balance D M 2 Errolex, Errolphyment expectations noted (NACE 64, 65, 66) D M 2 Errolex, Errolphyment expectations over the next 3 months, balance D M 2 Errolex, Errolphyment expectations over the next 3 months, balance D M 2 Errolex, Errolphyment expectations over the next 3 months, balance D M 2 Errolex, Errolphyment expectations over the next 3 months, balance D M 2 Errolex, Errolphyment expectat	Interest rates on loans of households (new business), consumer creat, notating rate up to 1 year initial itation	D	M	2
Interest rates on Joans of bouseholds (austanding amounts), inducting up 1 year         D         M         2           Loares to non-MHS (outstanding amounts), odomestic residents, annual growth rates         D         M         1           Deposits of non-MHS held with MHS (outstanding amounts), odomestic residents, annual growth rates         D         M         1           Deposits of non-MHS held with MHS (outstanding amounts), odomestic residents, annual growth rates         D         M         1           Loare-to-deposita rates: total along corporations per new loans (ferm)         D         M         3           Volumes of new loans to curo area non-financial corporations per new loans (ferm)         D         M         3           Volumes of new loans to curo area non-financial corporations per new loans (ferm)         D         M         3           Volumes of new loans to curo area households, pure new loans (ferm)         D         M         3           Volumes of new loans to curo area households (2015-100)         D         M         3           Economic Sentiment Indicator         D         M         2           Economic Sentiment Indicator (feonomics Research Centre)         D         M         2           Employment Expectations Indicator (feonomics Research Centre)         D         M         2           Services, Sendindene Indicator (NACE 5, 56	Interest rates on loans of households (new business), other lending, floating rate up to 1 year initial fixation	D	M	2
Loars to non-MFIs (ustanding amounts), obtal, armual growth rates         D         M         1           Deposits of non-MFIs held with MFIs (ustanding amounts), obtal, armual growth rates         D         M         1           Deposits of non-MFIs held with MFIs (ustanding amounts), obtal, armual growth rates         D         M         1           Loars-to-deposit ratio: total loars to total deposits         Volumes of new loars to curo area non-financial corporations & euro area households, pure new loans (fcm)         D         M         3           Volumes of new loars to curo area non-financial corporations & euro area households, pure new loans (fcm)         D         M         3           Volumes of new loans to curo area households, pure new loans (fcm)         D         M         3           Volumes of new loans to curo area households, pure new loans (fcm)         D         M         3           Consumer Price Index, 2015=100)         D         M         2         Employment Expectations Indicator         D         M         2           Enroleyoment Expectations Indicator         D         M         2         Escretices, Evolution of demand over the past 3 months, balance         D         M         2           Services, Sensements O business situation over the past 3 months, balance         D         M         2           Services, Foulition of demand over the past 3 months,	Interest rates on loans of households (outstanding amounts), maturity up to 1 year	D	Μ	2
Loars to non-MIPs lod with MIPs (outstanding amounts), domestic residents, annual growth ratesDM1Deposits of non-MIPs held with MIPs (outstanding amounts), domestic residents, annual growth ratesDM1Loan-to-deposits ratic total lospositsDM2Volumes of new loans to euro area non-financial corporations & euro area households, pure new loans (Enn)DM3Volumes of new loans to euro area non-financial corporations for house purchase (Enn)DM3Volumes of new loans to euro area households, pure new loans (Enn)DM3Volumes of new loans to euro area households, pure new loans for house purchase (Enn)DM3Tornonics feel ndes, (2015=100)DM32Econonic Sentiment Indicator (Econonics Research Centre)DM2Enployment Expectations Indicator (Econonics Research Centre)DM2Enployment Expectations Indicator or the past 3 months, balanceDM2Services, Schidhene IndicatorDM2Services, Endifience IndicatorDM2Services, Endifience Indicator or the next 3 months, balanceDM2Services, Endifience Indicator (NACE 46, 66, 66)DM2Services, Confidence Indicator (NACE 46, 66, 66)DM2Services, Confidence Indicator NACE 46, 66, 66)DM2Services, Confidence IndicatorDM2Services, Confidence IndicatorDM2Cons	Loans to non-MFIs (outstanding amounts), total, annual growth rates	D	М	1
Deposits of non-MFIs held with MFIs (outstanding amounts), total, annual growth rates         D         M         1           Loan-to-deposit ratio: total loans to total deposits         D         M         3           Volumes of new loans to euro area non-financial corporations & euro area households, pure new loans (Emn)         D         M           Volumes of new loans to euro area non-financial corporations & euro area households, pure new loans (Emn)         D         M           Volumes of new loans to euro area households, pure new loans (Enn)         D         M         3           Volumes of new loans to euro area households, pure new loans (Enn)         D         M         3           Consumer Price Index, (2015=100)         D         M         3           Econonic Sentiment Indicator (Economics Research Centre)         D         M         2           Enrophyment Expectations Indicator (Economics Research Centre)         D         M         2           Evervices, Sessements of business situation over the past 3 months, balance         D         M         2           Services, Fixeduition indicator (Sconomics Research Centre)         D         M         2           Services, Evolution of demand over the past 3 months, balance         D         M         2           Services, Evolution of employment evert anonth, balance         D         M <t< td=""><td>Loans to non-MFIs (outstanding amounts), domestic residents, annual growth rates</td><td>D</td><td>Μ</td><td>1</td></t<>	Loans to non-MFIs (outstanding amounts), domestic residents, annual growth rates	D	Μ	1
Deposits of non-MITs held with MIEs (outstanding amounts), domestic residents, annual growth rates         D         M           Lonar-to-deposit ratitis total loopsits         D         M         2           Volumes of new loans to euro area non-financial corporations & euro area households, pure new loans (Emn)         D         M           Volumes of new loans to euro area non-financial corporations for house purchase (Emn)         D         M           Volumes of new loans to euro area households, pure new loans (Enn)         D         M           Volumes of new loans to euro area households, pure new loans (Enn)         D         M           Tommorised Index of Consumer Prices (IIICP), (2015=100)         D         M           Consumer Friee Index, (2015=100)         D         M         2           Economic Sentiment Indicator         D         M         2           Economic Sentiment Indicator         D         M         2           Employment Expectations Indicator         D         M         2           Services, Sensemate of busines situation over the past 3 months, balance         D         M         2           Services, Evolution of demand over the past 3 months, balance         D         M         2           Services, Sengotyment expectations over the next 3 months, balance         D         M         2	Deposits of non-MFIs held with MFIs (outstanding amounts), total, annual growth rates	D	М	1
Lonk-odeposit Lonk-odeposit<	Deposits of non-MFIs held with MFIs (outstanding amounts), domestic residents, annual growth rates	D	M	1
Troumes of new forms or early area non-financial corporations as pure new loans (Emn)       D       M       3         Volumes of new loans to euro area non-financial corporations as pure new loans (Emn)       D       M       3         Volumes of new loans to euro area households, pure new loans (Enn)       D       M       3         Volumes of new loans to euro area households, pure new loans (Enn)       D       M       3         Consumer Price Index, (2015=100)       D       M       3         Consumer Their Indicator       D       M       2         Economic Sentiment Indicator       D       M       2         Enployment Expectations Indicator (Economics Research Centre)       D       M       2         Enrices, Assessments of business situation over the past 3 months, balance       D       M       2         Services, Colution of employment over the past 3 months, balance       D       M       2         Services, Fundoution of employment over the past 3 months, balance       D       M       2         Services, Fundoution of employment over the past 3 months, balance       D       M       2         Services, Fundoution of employment over the past 3 months, balance       D       M       2         Services, Financial Services Confidence Indicator (NACE 64, 65, 66)       D       M <t< td=""><td>Loan-to-deposit ratio: total loans to total deposits</td><td></td><td>M</td><td>2</td></t<>	Loan-to-deposit ratio: total loans to total deposits		M	2
Volumes of new loans to ear area households, pure new loans (ferm)       D       M       3         Volumes of new loans to ear area households, pure new loans for house purchase (ferm)       D       M       3         Volumes of new loans to ear area households, pure new loans for house purchase (ferm)       D       M       3         Consumer Price Index, (2015=100)       D       M       3         Economic Sentiment Indicator (Economics Research Centre)       D       M       2         Enployment Expectations Indicator (Economics Research Centre)       D       M       2         Services, Assessments of business situation over the past 3 months, balance       D       M       2         Services, Evolution of demand over the past 3 months, balance       D       M       2         Services, Evolution of employment over the past 3 months, balance       D       M       2         Services, Evolution of employment over the next 3 months, balance       D       M       2         Services, Capacity utilisation index       D       Q       1       2         Services, Capacity utilisation index       D       Q       1       2         Services, Capacity utilisation index       D       Q       1       2       2       2       2       2       2       2       2 <td>Volumes of new loans to euro area non-financial corporations, pure new loans (enrif)</td> <td>D</td> <td>M</td> <td>3</td>	Volumes of new loans to euro area non-financial corporations, pure new loans (enrif)	D	M	3
Volumes of new loans to euro area households, pure new loans for house purchase (fmn)DM3Consumer Price Index, (2015=100)DM3Cronomic Sentiment IndicatorDM3Economic Sentiment IndicatorDM2Employment Expectations IndicatorDM2Employment Expectations IndicatorDM2Services, Assessments of business situation over the past 3 months, balanceDM2Services, Condition of Montano Ver the past 3 months, balanceDM2Services, Conditions over the next 3 months, balanceDM2Services, Engloyment over the past 3 months, balanceDM2Services, Conditions over the next 3 months, balanceDM2Services, Fulling price expectations over the next 3 months, balanceDM2Services, Fulling price expectations over the next 3 months, balanceDM2Services, Fulling price expectations over the next 3 months, balanceDM2Services, Canacity utilisation indexDQ12Services, Canacity utilisation indexDM22Services, Canacity utilisation indexDM22Retail Trade, Assessments of busines activity (sales) over the past 3 months, balanceDM2Retail Trade, Intentions or placing orders over the next 3 months, balanceDM2Retail Trade, Intentions or placing orders over the next 3 months, balanceD </td <td>Volumes of new Joans to euro area households, pure new Joans (em)</td> <td>D</td> <td>M</td> <td>3</td>	Volumes of new Joans to euro area households, pure new Joans (em)	D	M	3
Harmonised Index of Consumer Prices (HICP), (2015=100)     D     M     3       Consumer Friee Index, 2015=100)     D     M     3       Price Index of construction materials, (2015=100)     D     M     3       Economic Sentiment Indicator     D     M     2       Eroployment Expectations Indicator     D     M     2       Employment Expectations Indicator     D     M     2       Services, Assessments of business situation over the past 3 months, balance     D     M     2       Services, Issessments of business situation over the past 3 months, balance     D     M     2       Services, Exolution of employment over the past 3 months, balance     D     M     2       Services, Fixolution of employment over the past 3 months, balance     D     M     2       Services, Fixolution of employment over the next 3 months, balance     D     M     2       Services, Fixolution of enployment expectations over the next 3 months, balance     D     M     2       Services, Capacity utilisation index     D     M     2       Services, Capacity utilisation index     D     M     2       Retail Trade, Assessments of business activity (sales) over the past 3 months, balance     D     M     2       Retail Trade, Sales expectations over the next 3 months, balance     D     M <t< td=""><td>Volumes of new loans to euro area households, pure new loans for house purchase (Emn)</td><td>D</td><td>M</td><td>3</td></t<>	Volumes of new loans to euro area households, pure new loans for house purchase (Emn)	D	M	3
Consumer Price Index, (2015–100)DM3Economic Sentiment IndicatorDM2Economic Sentiment Indicator (Economics Research Centre)DM2Employment Expectations IndicatorDM2Employment Expectations IndicatorDM2Enroloyment Expectations IndicatorDM2Services, Confidence IndicatorDM2Services, Notlidence IndicatorDM2Services, Deutiditon of employment over the past 3 months, balanceDM2Services, Evolution of demapment expectations over the next 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Fuelds & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Fuelds & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDM2Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2	Harmonised Index of Consumer Prices (HICP), (2015=100)	D	М	3
Price Index of construction materials, (2015–100)DM3Economic Sentiment IndicatorDM2Economic Sentiment Indicator (Economics Research Centre)DM2Employment Expectations Indicator (Economics Research Centre)DM2Services, Confidence IndicatorDM2Services, Sessements of business situation over the past 3 months, balanceDM2Services, Evolution of demand over the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the next 3 months, balanceDM2Services, Financial Services Confidence Indicator (NACE 55, 56)DM2Services, Financial Services Confidence Indicator (NACE 55, 56)DM2Services, Financial Services Confidence Indicator (NACE 55, 56)DM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM2Retail Trade,	Consumer Price Index, (2015=100)	D	Μ	3
Economic Sentiment IndicatorDM2Economic Sentiment Indicator (Economics Research Centre)DM2Employment Expectations Indicator (Economics Research Centre)DM2Services Confidence IndicatorDM2Services Confidence IndicatorDM2Services, Assessments of business situation over the past 3 months, balanceDM2Services, Foolution of demand over the past 3 months, balanceDM2Services, Foolution of employment over the past 3 months, balanceDM2Services, Foolution of employment over the next 3 months, balanceDM2Services, Foolution of employment over the next 3 months, balanceDM2Services, Fotelis & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDQ1Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Seles expectations over the next 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Construction, Assessments of ourient expectations over the next 3 months, balanceDM2<	Price Index of construction materials, (2015=100)	D	Μ	3
LCOODDITYDM2Employment Expectations IndicatorDM2Employment Expectations Indicator (Economics Research Centre)DM2Services, Confidence IndicatorDM2Services, Assessments of business situation over the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the next 3 months, balanceDM2Services, Services, Services, Services, Capacity utilisation indexDM2Services, Fronkis & Research Control, SolanceDM2Services, Fronkis & Research Control, SolanceDM2Services, Choles & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDQ1Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Selessments of ourtent over the next 3 months, balanceDM2Construction, Selling price ex	Economic Sentiment Indicator	D	M	2
Imployment Expectations Indicator (Economics Research Centre)DM2Employment Expectations Indicator (Economics Research Centre)DM2Services, Assessments of business situation over the past 3 months, balanceDM2Services, Dessessments of vour the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution over the next 3 months, balanceDM2Services, Evolution over the next 3 months, balanceDM2Services, Folling price expectations over the next 3 monthsDM2Services, Foldes & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDQ11Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Selse expectations over the next 3 months, balanceDM2Retail Trade, Selse expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction Confidence IndicatorDM2Construction, Assessments of vulting activity over the past 3 months, balanceDM2Construction, Cassessments of vulting activity over the past 3 months, balanceDM2Construction, Selsengent of busiling activity over the past 3 months, ba	Economic Sentiment indicator (Economics Research Centre)		M	2
InstructionDM2Services Confidence IndicatorDM2Services, Devolution of demand over the past 3 months, balanceDM2Services, Devolution of demand over the past 3 months, balanceDM2Services, Devolution of employment over the past 3 months, balanceDM2Services, Evolution of employment expectations over the next 3 months, balanceDM2Services, Volution of employment expectations over the next 3 months, balanceDM2Services, Fourbis & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Fourbis & Restaurants Confidence Indicator (NACE 54, 65, 66)DM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of volume of stock currently hold, balanceDM2Construction, Assessments of current over the next 3 months, balanceDM2Construction, Confidence IndicatorDM22Construction, Assessments of current over all order books, balanceDM2Construction, Selfurp price expectations over the next 3 months, balanceDM2	Employment Expectations Indicator (Economics Research Centre)	D	M	2
Services, Assessments of business situation over the past 3 months, balanceDM2Services, Evolution of demand over the past 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the next 3 months, balanceDM2Services, Employment expectations over the next 3 monthsDM2Services, Functiona Sore the next 3 monthsDM2Services, Functional Services Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDQ1Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Assessments of builing activity over the past 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2 <td< td=""><td>Services Confidence Indicator</td><td>D</td><td>M</td><td>2</td></td<>	Services Confidence Indicator	D	M	2
Services, Evolution of demand over the past 3 months, balanceDM2Services, Demand expectations over the next 3 months, balanceDM2Services, Evolution of employment over the past 3 months, balanceDM2Services, Evolution of employment over the next 3 months, balanceDM2Services, Evolution of employment over the next 3 monthsDM2Services, Financial Services Confidence Indicator (NACE 55, 56)DM2Services, Financial Services Confidence Indicator (NACE 64, 65, 66)DM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Easles expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM22Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Saling price expectations over the next 3 months, balanceDM2Construction, Sessments of current overall order books, balanceDM2Construction, Saling price e	Services, Assessments of business situation over the past 3 months, balance	D	М	2
Services, Demand expectations over the next 3 months, balanceDM2Services, Einployment expectations over the next 3 months, balanceDM2Services, Fubles & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Fubles & Restaurants Confidence Indicator (NACE 64, 65, 66)DM2Services, Fubles & Restaurants Confidence Indicator (NACE 64, 65, 66)DM2Services, Capacity utilisation indexDQ1Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM2Construction, Confidence IndicatorDM2Construction, Confidence IndicatorDM2Construction, Assessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Sessesments of current versall order books, balanceDM2Construction, Sessesments of current versall order books, balanceDM2Construction, Sessesment of current expectations over the next 3 months,	Services, Evolution of demand over the past 3 months, balance	D	Μ	2
Services, Evolution of employment over the past 3 months, balanceDM2Services, Employment expectations over the next 3 months, balanceDM2Services, Foltels & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Financial Services Confidence Indicator (NACE 64, 65, 66)DM2Services, Capacity utilisation indexDQ1Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM2Construction, Assessments of current overall order books, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2<	Services, Demand expectations over the next 3 months, balance	D	М	2
Services, Employment expectations over the next 3 monthsDefaultM2Services, Selling price expectations over the next 3 monthsDM2Services, Financial Services Confidence Indicator (NACE 55, 56)DM2Services, Capacity utilisation indexDQ1Retail Trade Confidence Indicator (NACE 64, 65, 66)DM2Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Selie seprectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Sessesments of production over the next 3 months, balanceDM2Construction, Geneting time ensured by current backlog, monthsDM2Construction, Sessesment of production over the next 3 months, balanceDM2Industry, Assessment of current texclo of finished products, balanceDM2Industry, Confidence Indicator <t< td=""><td>Services, Evolution of employment over the past 3 months, balance</td><td>D</td><td>M</td><td>2</td></t<>	Services, Evolution of employment over the past 3 months, balance	D	M	2
Services, Hotels & Restaurants Confidence Indicator (NACE 55, 56)DM2Services, Friancial Services Confidence Indicator (NACE 64, 65, 66)DQ1Retail Trade Confidence IndicatorDM2Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Seling price expectations over the next 3 months, balanceDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3	Services, Employment expectations over the next 3 months, balance	D	M	2
Services, Financial Services Confidence Indicator (NACE 64, 65, 66)DM2Services, Capacity utilisation indexDQ1Retail Trade Confidence IndicatorDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Selse expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM2Industry, Assessment of current over 10 acdr books, balanceDM2Industry, Assessment of current sport order books, balanceDM2Industry, Assessment of current stock of finished products, balance <t< td=""><td>Services, Sening price expectations over the next 5 months</td><td>D</td><td>M</td><td>2</td></t<>	Services, Sening price expectations over the next 5 months	D	M	2
Services, Capacity utilisation indexDQ1Retail Trade Confidence IndicatorDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Early construction control the next 3 months, balanceDM2Retail Trade, Early construction Confidence IndicatorDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDQ1Industry, Assessment of current over the past 3 months, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current backlog, monthsDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current export order books, balanceDM2 </td <td>Services, Financial Services Confidence Indicator (NACE 64, 65, 66)</td> <td>D</td> <td>M</td> <td>2</td>	Services, Financial Services Confidence Indicator (NACE 64, 65, 66)	D	M	2
Retail Trade Confidence IndicatorDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Saessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Sessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Confidence IndicatorDM2Industry Confidence IndicatorDM2Industry, Assessment of current backlog, monthsDM2Industry, Assessment of current export order books, balanceDM2Industry, Sellin	Services, Capacity utilisation index	D	Q	1
Retail Trade, Assessments of business activity (sales) over the past 3 months, balanceDM2Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDM2Construction, Operating time ensured by current backlog, monthsDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Production expectati	Retail Trade Confidence Indicator	D	М	2
Retail Trade, Assessments of volume of stock currently hold, balanceDM2Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Seessments of current overall order books, balanceDM2Construction, Seessments of price expectations over the next 3 months, balanceDM2Construction, Seessments of current overall order books, balanceDM2Construction, Seessment of current overall order books, balanceDM2Construction, Seessment of production over the next 3 months, balanceDM2Industry Confidence IndicatorDM2Industry, Assessment of current backlog, monthsDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current expect order books, balanceDM2Industry, Assessment of current expect order books, balanceDM2Industry, Assessment of current expect order books, balanceDM2Industry, Selling price expectations for the next 3 months, ba	Retail Trade, Assessments of business activity (sales) over the past 3 months, balance	D	Μ	2
Retail Trade, Intentions of placing orders over the next 3 months, balanceDM2Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Retail Trade, Selling price expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Confidence IndicatorDM2Construction, Operating time ensured by current backlog, monthsDM2Construction, Operating time ensured by current backlog, monthsDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Selling price expectations over the next 3 months, balanceDM2Industry, Semployment expectations for the next 3 months, balanceDM <td>Retail Trade, Assessments of volume of stock currently hold, balance</td> <td>D</td> <td>Μ</td> <td>2</td>	Retail Trade, Assessments of volume of stock currently hold, balance	D	Μ	2
Retail Trade, Sales expectations over the next 3 months, balanceDM2Retail Trade, Employment expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDM2Construction, Operating time ensured by current backlog, monthsDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Cassessment of current stock of finished products, balanceDM2Industry, Cassessment of current stock of finished products, balanceDM2Industry, Forduction expectations over the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Employment expectations for the	Retail Trade, Intentions of placing orders over the next 3 months, balance	D	M	2
Retail Trade, Enlipsyment expectations over the next 3 months, balanceDM2Construction Confidence IndicatorDM2Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDM2Construction, Operating time ensured by current backlog, monthsDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM21Consumer Confidence IndicatorDM </td <td>Retail Trade, Sales expectations over the next 3 months, balance</td> <td></td> <td>M</td> <td>2</td>	Retail Trade, Sales expectations over the next 3 months, balance		M	2
Initial Tuber Construction Confidence IndicatorDM2Construction Confidence IndicatorDM2Construction, Assessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDM2Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Seessment of current stock of finished products, balanceDM2Industry, Seessment of current stock of finished products, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General ec	Retail Trade Selling price expectations over the next 3 months balance	D	M	2
Construction, Assessments of building activity over the past 3 months, balanceDM2Construction, Assessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDQ1Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Seessment of current export order books, balanceDM2Industry, Seling price expectations for the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the	Construction Confidence Indicator	D	M	2
Construction, Assessments of current overall order books, balanceDM2Construction, Employment expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDQ1Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Industry, Capacity utilisation indexDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, bala	Construction, Assessments of building activity over the past 3 months, balance	D	М	2
Construction, Employment expectations over the next 3 months, balanceDM2Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDQ1Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Industry, Capacity utilisation indexDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months,	Construction, Assessments of current overall order books, balance	D	Μ	2
Construction, Selling price expectations over the next 3 months, balanceDM2Construction, Operating time ensured by current backlog, monthsDQ1Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Industry, Capacity utilisation of household over the last 12 months, balanceDM2Consumers, Financial situation of household over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Construction, Employment expectations over the next 3 months, balance	D	Μ	2
Construction, Operating time ensured by current backlog, monthsDQIIndustry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Construction, Selling price expectations over the next 3 months, balance	D	M	2
Industry Confidence IndicatorDM2Industry, Assessment of production over the past 3 months, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Lonstruction, Operating time ensured by current backlog, months	D	Q	1
Industry, Assessment of current order books, balanceDM2Industry, Assessment of current order books, balanceDM2Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2	Industry Connuence indicator	D	M	2
Industry, Assessment of current export order books, balanceDM2Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2Onsumers, General economic situation over the next 12 months, balanceDM2	Industry, Assessment of current order books, balance	D	M	2
Industry, Assessment of current stock of finished products, balanceDM2Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDM2Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Industry, Assessment of current export order books, balance	D	Μ	2
Industry, Production expectations over the next 3 months, balanceDM2Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Industry, Assessment of current stock of finished products, balance	D	М	2
Industry, Selling price expectations for the next 3 months, balanceDM2Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Industry, Production expectations over the next 3 months, balance	D	Μ	2
Industry, Employment expectations for the next 3 months, balanceDM2Industry, Capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, Financial situation of household over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Industry, Selling price expectations for the next 3 months, balance	D	М	2
Industry, capacity utilisation indexDQ1Consumer Confidence IndicatorDM2Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, General economic situation over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Industry, Employment expectations for the next 3 months, balance	D	M	2
Consumers, Financial situation of household over the last 12 months, balanceDM2Consumers, Financial situation of household over the next 12 months, balanceDM2Consumers, General economic situation over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	industry, Capacity utilisation index Consumer Confidence Indicator	ע ת	Q	1
Consumers, Financial situation of household over the next 12 months, balanceDM2Consumers, General economic situation over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Consumers Financial situation of household over the last 12 months balance	ם ח	M	∠ 2
Consumers, General economic situation over the last 12 months, balanceDM2Consumers, General economic situation over the next 12 months, balanceDM2	Consumers, Financial situation of household over the next 12 months, balance	D	M	2
Consumers, General economic situation over the next 12 months, balance D M 2	Consumers, General economic situation over the last 12 months, balance	D	М	2
	Consumers, General economic situation over the next 12 months, balance	D	М	2

# TABLE A2 (continued)

A         A           Consumes, Price trends over the test 12 months, balance         D           Consumes, Price trends over the next 12 months, balance         D           Consumers, Major purchase apprent, balance         D           Consumers, Swings over the next 12 months, balance         D           Consumers, Swings over the next 12 months, balance         D           Consumers, Swings over the next 12 months, balance         D           Consumers, Swings over the next 12 months, balance         D           Consumers, Furchase or build a home within the next 12 months, balance         D           Consumers, Furchase or build a home within the next 12 months, balance         D           Consumers, Furchase or build a home within the next 12 months, balance         D           Consumers, Nachase or build a home within the next 12 months, balance         D           Uncertainty indicator, ex opectations (row (2015-100)         F           FL3V, Volume index of manufacturing production, (2015-100)         F           FL3V, Solume index of manufacturing production, (2015-100)         F           FL3V, Rebail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015-100)         F           FL4N, Rebail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015-100)         F           FL4N, Rebail - except of motor vehicle and mototrexcles, Index of deflated turnove	B         C           M         2           M         2           M         2           M         2           M         2           M         2           M         2           M         2           M         2           Q         2           Q         2           Q         2           M         1           M         3           M         3           M         3           Q         1
Consumes, Price trends over the last 12 months, balance         D           Consumes, Unemployment expectations over the next 12 months, balance         D           Consumers, Major purchases are present, balance         D           Consumers, Major purchases are present, balance         D           Consumers, Savings avore the next 12 months, balance         D           Consumers, Savings over the next 12 months, balance         D           Consumers, Savings over the next 12 months, balance         D           Consumers, Functions to buy a car within the next 12 months, balance         D           Consumers, Functiacor to buy a car within the next 12 months, balance         D           Consumers, Functiacor evalue and none within the next 12 months, balance         D           Consumers, Functiacor evalue and none vergiture next 12 months, balance         D           Uncertainty indicator, ex opet, expeciation errors (business surveys)         D           Uncertainty indicator, ex opet and motor velice and motorcycles, Index of deflated turnover, (2015–100)         F           EA3, Reital CP growth expectations, current calendar year (ECB Survey of Professional Forecasters)         F           EA, Reital CP growth expectations, current calendar year (ECB Survey of Professional Forecasters)         F           EA, Unemployment rate         F           EA, Unemployment rate         F           EA, Unemplo	M 2 M 2 M 2 M 2 M 2 M 2 M 2 M 2 M 2 M 2
Consumes, Virce tends over the next 12 months, balance D Consumes, Major purchases or the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Savings over the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance D Consumes, Furchase or build a home within the next 12 months, balance C D Consumes, Furchase or build a montorycles, Index of delated turnover, (2015–100) F EU27, Retail – except of motor vehicle and motorycles, Index of delated turnover, (2015–100) F EA, Red ICDP growth expectations, next calendar year (ECB Survey of Professional Forecasters) F EU, Lonenployment rate EF EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (ECB Survey of Professional Forecasters) F EA, Lonenployment rate (EC	M 2 M 2 M 2 M 2 M 2 M 2 M 2 M 2 Q 2 Q 2 Q 2 Q 2 Q 2 Q 2 M 1 M 3 M 3 M 3 M 3 Q 1
Consumers, Major purchases at present, balance in home bounded of a set of the set of home bound of the set of the set of home bound of the set of the set of home bound of the set of the set of home bound of the set of home bound of home bo	M 2 M 2 M 2 M 2 M 2 Q 2 Q 2 Q 2 Q 2 Q 2 Q 2 M 1 M 3 M 3 M 3 M 3 Q 1
Consumers, Major purchases over the next 12 months, balance     D       Consumers, Swings at present, balance     D       Consumers, Swings at present, balance     D       Consumers, Turchase or build a home within the next 12 months, balance     D       Consumers, Turchase or build a home within the next 12 months, balance     D       Consumers, Turchase or build a home within the next 12 months, balance     D       Consumers, Turchase or build a home within the next 12 months, balance     D       Uncertainty indicator, ex ante employment and price expectations (business surveys)     D       Uncertainty indicator, ex opet, expectation (cultisens surveys)     D       EUZ, Volume index of manufacturing production, (2015-100)     F       EUZ, Nolume index of manufacturing production, (2015-100)     F       ELAB, Real I-DE growth expectations, next calendar year (ECB Survey of Professional Forecasters)     F       EU, Lonenployment rate     F       EU, Chemployment rate     F       EA, Neal CDP growth expectations, next calendar year (ECB Survey of Professional Forecasters)     F       EU, Chemployment rate     F       EA, Unemployment rate     F       EA, Unemployment rate     F       Europe, Gennth EURIBOR     F       Europe Associations, next calendar year (ECB Survey of Professional Forecasters)     F       Europe Chennth EURIBOR     F       Europe Associ	M       2         M       2         M       2         Q       2         Q       2         Q       2         Q       2         M       1         M       3         M       3         M       3         Q       1         Q       1
Consumers, Savings our the next 12 months, balance     D       Consumers, Statement on financial situation of household, balance     D       Consumers, Funchios to bry a car within the next 12 months, balance     D       Consumers, Humei improvements over the next 12 months, balance     D       Consumers, Funchase or build a home within the next 12 months, balance     D       Uncertainty indicator, ex ante employment and price expectations (business surveys)     D       Uncertainty indicator, ex post, expectation errors (business surveys)     D       Uncertainty indicator, ex post, expectation errors (business surveys)     D       EX17, Volume index of manufacturing production, (2015–100)     F       EA19, Volume index of manufacturing production, (2015–100)     F       EA, Real CDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)     F       EA, Real CDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)     F       EA, Unemployment rate     F       EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)     F       Europe, Carnonth EURIBOR     F       Europe, I2-month EURIBOR     F	M 2 M 2 Q 2 Q 2 Q 2 Q 2 M 1 M 1 M 3 M 3 M 3 M 3 Q 1
Consumers, Savings over the next 12 months, balance     D       Consumers, Intention to buy a car within the next 12 months, balance     D       Consumers, Intention to buy a car within the next 12 months, balance     D       Consumers, Home improvements over the next 12 months, balance     D       Consumers, Home improvements over the next 12 months, balance     D       Uncertainty indicator, ex and temployment and price expectations (business surveys)     D       Uncertainty indicator, ex post, expectation errors (business surveys)     D       Uzz, Volume index of manufacturing production, (2015–100)     F       EA19, Notami endex of manufacturing production, (2015–100)     F       EA19, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015–100)     F       EA, Real GDP growth expectations, next calendar year (ECB Survey of Professional Forecasters)     F       EU, Inemployment rate     F       EV, Unemployment rate     F       EA, Unemployment rate     F       EA, Unemployment rate     F       Europe, S-month EURIBOR     F       Europe, Carmonth EURIBOR     F       Europe, Carmonth EURIBOR     F       Europe accentral Bank deposit rate     F       Europe	M 2 M 2 Q 2 Q 2 Q 2 M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
Consumers, Statement on financial situation of household, balance D Consumers, Purchase or build a home within the next 12 months, balance D Consumers, Purchase or build a home within the next 12 months, balance D Uncertainty indicator, ex ante employment and price expectations (business surveys) D Uncertainty indicator, ex ante, expectation records (business surveys) D EU22, Volume index of manufacturing production, (2015=100) F EU23, Retail - except of motor vehicle and motorycles, Index of deflated turnover, (2015=100) F EA19, Volume index of manufacturing production, (2015=100) F EU24, Retail - except of motor vehicle and motorycles, Index of deflated turnover, (2015=100) F EA, Real GDP growth expectations, current clandar year (ECB Survey of Professional Forecasters) F EA, Real GDP growth expectations, current clandar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate EA, Unemployment PA, EA, EA, EA, EA, EA, EA, EA, EA, EA, E	M 2 Q 2 Q 2 Q 2 M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
Consumes, Function to buy a car writing the next 12 months, balance D Consumers, Home improvements over the next 12 months, balance D Consumers, Home improvements over the next 12 months, balance D Uncertainty indicator, ex post, expectation errors (business surveys) D Uncertainty indicator, ex post, expectation errors (business surveys) D EX19, Volume index of manufacturing production, (2015–100) F EX19, Volume index of manufacturing production, (2015–100) F EX19, Volume index of motor vehicle and motorcycles, Index of deflated turnover, (2015–100) F EX19, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015–100) F EX19, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015–100) F EX19, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015–100) F EX, Retail OF growth expectations, next calendar year (ECB Survey of Professional Forecasters) F EX, Unemployment rate EX, Unemployment rate EX, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F Europe, 3-month EURIBOR Europe and the expectations, next calendar year (ECB Survey of Professional Forecasters) F Europe, 3-month EURIBOR Europe and the expectations, next calendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe and the expectations, next calendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe and the expectations, next calendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe and the expectations, euror talendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe and the expectations, euror talendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe anoth EURIBOR Europe and the expectations, euror talendar year (ECB Survey of Professional Forecasters) F Europe anoth EURIBOR Europe anoth EURIBOR Europe and Covernment Bond Yield (DE10) F Germany. 3-month Treasury Bill Yield F Spain, 3-pear Gove	Q 2 Q 2 M 1 M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
Consumes, Home improvements over the next 12 months, balanceDConsumes, Home improvements over the next 12 months, balanceDUncertainty indicator, ex ante employment and price expectations (business surveys)DEUZ, Volume index of manufacturing production, (2015=100)FEUZ, Volume index of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)FEUZ, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)FEA, Real GDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)FEA, Real GDP growth expectations, next calendar year (ECB Survey of Professional Forecasters)FEA, Unemployment rateFEA, Unemployment rateFEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEarope, 3-month EURIBORFEurope, 3-month EURIBORFEurope, 12-month EURIBORFEuropean Central Bank deposit rateFEuropean Covernment Benchmark Bond Yield (DE10)FGermany, 10-year Government Benchmark Bond Yield (DE10)FFrance, 10-year Government Benchmark Bond Yield (EIS10)FSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill Yield (EIS10)FSpain, 3-month Treasury Bill Yield (EIS10)FSpain, 3-month Treasury Bill Yield (EIS10)FSpain, 3-month Treasury Bill Yield (EIS10)FSpread STR0 and DE10FSpread STR0 and DE10F	Q 2 M 1 M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
Uncertainty indicator, ex ante employment and price expectations (business surveys)DUncertainty indicator, ex post, expectation errors (business surveys)DEX27, Volume index of manufacturing production, (2015–100)FEA19, Volume index of manufacturing production, (2015–100)FEA19, Volume index of manufacturing production, (2015–100)FEA19, Except of motor vehicle and motorcycles, Index of deflated turnover, (2015–100)FEA, Real CDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)FEA, Lonemployment rateFEA, Unemployment rateFEA, Unemployment rateFEA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)FEurope, 3-month EURIBORFEurope, 3-month EURIBORFEurope, 12-month EURIBORFEurope Carrent Bank deposit rateFEurope Comoth FURIBORFEurope Comoth FURIBORFEurope Comoth Fuestions, Europe (ECB UP)FEurope Comoth EURIBORFEurope Comoth EURIBORFEurope Comoth EURIBORFEurope Comoth FURIBORFEurope Comoth FURIBORFEurope Comoth Frasury Bill YieldFFrance, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFSpread EXI0 and DEI0FSpread EX	M 1 M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
Uncertainty indicator, ex post, expectation errors (business surveys)DUDFEU27, Volume index of manufacturing production, (2015=100)FEV19, Vetail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)FEV19, Ketail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)FEA19, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)FEA, Real GDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)FEU, Unemployment rateFEU, Unemployment rateFEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEurope, 6-month EURIBORFEurope, 6-month EURIBORFEurope, 6-month EURIBORFEurope, 2-month EURIBORFEurope, 2-month EURIBORFEurope, 2-month EURIBORFEurope, 2-month EURIBORFEurope, 2-month EURIBORFEuropean Central Bank deposit rateFEuropean Contral Bank deposit rateFEuropean Covernment Benchmark Bond Yield (DE10)FFrance, 10-year Government Bond Yield (FR10)FFrance, 10-year Government Bond Yield (FR10)FSpin, 3-month Treasury Bill YieldFSpin, 3-month Treasury Bill YieldFSpin, 3-month Treasury Bill YieldFSpin, 3-month Treasury Bill YieldF<	M 1 M 3 M 3 M 3 M 3 Q 1 Q 1
EU27, Volume index of manufacturing production, (2015=100)       F         EV19, Volume index of manufacturing production, (2015=100)       F         EV27, Retail - except of motor vehicle and motorcycles, Index of deflated turnover, (2015=100)       F         EA, Real CDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)       F         EA, Real CDP growth expectations, next calendar year (ECB Survey of Professional Forecasters)       F         EA, Unemployment rate       F         EA, Unemployment rate       F         EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)       F         EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)       F         Europe, 3-month EURIBOR       F         Europe, 3-month EURIBOR       F         Europe, 12-month EURIBOR       F         European Central Bank Inding rate       F         European Central Bank Inding rate       F         European Central Bank Inding rate       F         France, 10-year Government Bond Yield (DE10)       F         Germany, 10-year Government Bond Yield (FT10)       F         Ialay, 10-year Government Bond Yield (ES10)       F         Spain, 10-year Government Bond Yield (ES10)       F         Spain, 10-year Government Bond Yield (ES10)	M 3 M 3 M 3 M 3 Q 1 Q 1
EA19, Volume index of manufacturing production, (2015=100) F EA19, Retail - except of motor vehicle and motorscycles, Index of deflated turnover, (2015=100) F EA, Real GDP growth expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Real GDP growth expectations, next calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate F EA, Unemployment rate F EA, Unemployment rate E EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F Europe, 6-month EURIBOR F Europea, 6-month EURIBOR F European Central Bank lending rate F European Central Bank lending rate F European Central Bank lending rate F Germany, 10-year Government Benchmark Bond Yield (DE10) F Germany, 10-year Government Benchmark Bond Yield (DE10) F France, 10-year Government Bond Yield (FR10) F France, 10-year Government Bond Yield (FR10) F France, 10-year Government Bond Yield (ES10) F Spain, Jononth Treasury Bill Yield F Fandy, Joyear Government Bond Yield (ES10) F Spain, Joyear Government Bond Yield (ES10) F Spain, Joyear Government Bond Yield (US10) F K13, 3-month Treasury Bill Yield F K143, 3-month Treasury Bill Yield F Furo Duv Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Duv Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Duv Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Duv Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Duv Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Duv Jones Euro Stoxx 50 Price Index, Euro area (changing comp	M 3 M 3 M 3 Q 1 Q 1
BL27, Netali - Except or motor velicle and motorcycles, index of deflated turnover, (2015–100)       F         EA, Real GDP growth expectations, current calendar year (ECB Survey of Professional Forecasters)       F         EA, Real GDP growth expectations, next calendar year (ECB Survey of Professional Forecasters)       F         EU, Unemployment rate       F         EA, Unemployment rate       F         EA, Unemployment rate       F         EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)       F         EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)       F         Europe, 3-month EURIBOR       F         Europe, 3-month EURIBOR       F         Europe, 12-month EURIBOR       F         Europe, 12-month EURIBOR       F         Europe and Central Bank deposit rate       F         Europe anoth Treasury Bill Yield       F         France, 3-month Treasury Bill Yield       F         Spain, 10-year Government Benchmark Bond Yield (ES10)	M 3 Q 1 Q 1
EACS, Real CDP growth expectations, current calendar year (ECB Survey of Professional Forecasters) EA, Real CDP growth expectations, next calendar year (ECB Survey of Professional Forecasters) EA, Lonemployment rate EA, Unemployment rate EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) Europe, 3-month EURIBOR Europe, 6-month EURIBOR Europe, 6-month EURIBOR Europe Cannoth EURIBOR EURO ECON EUROPE EUROPE EUROPE EUROPE Euro David Covernment Bond Yield (ERI0) FORECC, 10-year Government Bond Yield (EII0) FORECC, 3-month Treasury Bill Yield FORECC ECON EUROPE EUROPE EUROPE Euro David Covernment Bond Yield (UKI0) EUR, 3-month Treasury Bill Yield FORECC EDV EUROPE EUROPE EURO DAV Jones Euro Stoxx 50 Price Index, Euro area (changing composition) FORECC EDVE EUROPE EUROPE EUROPE EURO DAV JONES EURO STOXX Frice Index, Euro area (changing composition) FORECC EDVE EUROPE EUROPE EURO	Q 1 Q 1
EA, Real GDP growth expectations, next calendar year (ECB Survey of Professional Forecasters) F EU, Unemployment rate F EU, Unemployment rate EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters) F EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters) F Earope, 5-month EURIBOR F Europe, 6-month EURIBOR F Europe, 6-month EURIBOR F European Central Bank deposit rate European Central Bank lending rate Germany, 0-year Government Benchmark Bond Yield (DE10) F France, 1-year Government Benchmark Bond Yield (DE10) F France, 1-year Government Benchmark Bond Yield (TI0) Italy, 3-month Treasury Bill Yield F Spain, 10-year Government Benchmark Bond Yield (ES10) F Spain, 3-month Treasury Bill Yield F Spain, 10-year Government Note Yield (EL10) F Greece, 0-year Government Note Yield (EL10) F Greece, 0-year Government Note Yield (UK10) F Spread ES10 and DE10 F Spread ES10	$\tilde{Q}$ 1
EU, Unemployment rateFEA, Unemployment rateFEA, Unemployment rateFRussia, Unemployment rateFEA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)FEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEurope, 3-month EURIBORFEurope, 3-month EURIBORFEuropean Central Bank leposit rateFEuropean Central Bank lending rateFGermany, 10-year Government Benchmark Bond Yield (DE10)FGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Benchmark Bond Yield (T10)FItaly, 10-year Government Benchmark Bond Yield (T10)FItaly, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFFrance, 10-year Government Net Yield (EL10)FGreece, 3-month Treasury Bill YieldFK1, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFFrance, 10-year Government Net Yield (EL10)FGreece, 10-year Government Bond Yield (UK10)FUK, 10-year Government Bond Yield (UK10)FEuropean Generic Both YieldFSpread FEI and DE10FSpread FEI and DE10FSpread ES10 and DE10FFur Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFur Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFur Dow Jones Euro S	-
EA, Unemployment rateFUK, Unemployment rateFRussia, Unemployment rateFEA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)FEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEurope, 3-month EURIBORFEurope, 6-month EURIBORFEurope, 6-month EURIBORFEurope, 6-month EURIBORFEuropean Central Bank deposit rateFEuropean Central Bank lending rateFGermany, 10-year Government Benchmark Bond Yield (DE10)FGermany, 10-year Government Benchmark Bond Yield (T10)FFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFysin, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Note Yield (EL10)FGreece, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpread EI10 and DE10FSpread EI10 and DE10FSpread EI10 and DE10FSpread EI10 and DE10FSpread EI10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEur	M 2
UK, Unemployment rateFRussia, Unemployment rateFEA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)FEA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)FEurope, Amonth EURIBORFEurope, 4-month EURIBORFEurope, 2-month EURIBORFEuropean Central Bank deposit rateFEuropean Central Bank lending rateFGermany, 10-year Government Benchmark Bond Yield (DE10)FFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFFlatly, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpreed FI10 and DE10FSpread FI10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)F <td>M 2</td>	M 2
Russia, Unemployment rate       F         EA, Unemployment rate expectations, current calendar year (ECB Survey of Professional Forecasters)       F         EA, Unemployment rate expectations, next calendar year (ECB Survey of Professional Forecasters)       F         Europe, 3-month EURIBOR       F         Europe, 2-month EURIBOR       F         Europe, 12-month EURIBOR       F         European Central Bank deposit rate       F         European Central Bank lending rate       F         Germany, 10-year Government Benchmark Bond Yield (DE10)       F         Germany, 3-month Treasury Bill Yield       F         France, 10-year Government Bond Yield (T10)       F         Italy, 10-year Government Benchmark Bond Yield (T10)       F         Italy, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 3-month Treasury Bill Yield       F         Spain, 3-month Treasury Bill Yield       F         Spain, 3-month Treasury Bill Yield       F         Spain, 10-year Government Note Yield (EL10)       F         Greece, 10-year Government Note Yield (UK10)       F         UK, 3-month Treasury Bill Yield       F         Spread ER10 and DE10       F         Spread ER10 and DE10 <td< td=""><td>M 2</td></td<>	M 2
EA, Unemployment rate expectations, current calendar year (ECB survey of Professional Forecasters)       F         EA, Unemployment rate expectations, next calendar year (ECB survey of Professional Forecasters)       F         Europe, 3-month EURIBOR       F         Europe, 6-month EURIBOR       F         Europe, 2-month EURIBOR       F         Europe and Central Bank lending rate       F         Germany, 10-year Government Benchmark Bond Yield (DE10)       F         Germany, 10-year Government Benchmark Bond Yield (T10)       F         France, 10-year Government Benchmark Bond Yield (T10)       F         Italy, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 3-month Treasury Bill Yield       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Note Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spain, 10-year Government Benchmark Bond Yield (ES10)       F         Spread FR10 and DE10       F         Spread FR10 and DE10       F         Spread FR10 and DE10	M 2
Inc. Other provide and expectations, next calendar year (acco Survey of Frotessional Forecasters)FEurope, 6-month EURIBORFEurope, 12-month EURIBORFEuropean Central Bank deposit rateFEuropean Central Bank lending rateFGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Benchmark Bond Yield (DE10)FFrance, 3-month Treasury Bill YieldFFrance, 3-month Treasury Bill YieldFItaly, 10-year Government Benchmark Bond Yield (IT10)FItaly, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Benchmark Bond Yield (ES10)FGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFGreece, 3-month Treasury Bill YieldFSpread FI10 and DE10FSpread FI10 and DE10FSpread EL10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)<	$\begin{array}{cc} 0 & 1\\ 0 & 1 \end{array}$
Europe, 6-month EURIBOR F Europe, 2-month EURIBOR F European Central Bank leposit rate F European Central Bank lending rate F Germany, 10-year Government Benchmark Bond Yield (DE10) F Germany, 3-month Treasury Bill Yield F France, 10-year Government Benchmark Bond Yield (DE10) F France, 3-month Treasury Bill Yield F France, 10-year Government Benchmark Bond Yield (IT10) F Italy, 10-year Government Benchmark Bond Yield (ES10) F Spain, 10-year Government Benchmark Bond Yield (ES10) F Spain, 3-month Treasury Bill Yield F Spain, 3-month Treasury Bill Yield F Greece, 10-year Government Note Yield (EL10) F Greece, 3-month Treasury Bill Yield F UK, 10-year Government Note Yield (EL10) F Greece, 3-month Treasury Bill Yield F UK, 10-year Government Bond Yield (UK10) F Spread FR10 and DE10 F Spread FR10 and DE10 F Spread EL10 Dato Stoxx 50 Price Index, Euro area (changing composition) F Euro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Streat DE10 Frie Index F CAC 40 Price Index F ATHEX Composite Price Index F ATHEX Composite Price Index F MTHEX Composite Price I	Q 1 M 2
Europe, 12-month EURIBORFEuropean Central Bank deposit rateFEuropean Central Bank deposit rateFGermany, 10-year Government Benchmark Bond Yield (DE10)FGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Benchmark Bond Yield (IT10)FItaly, 10-year Government Benchmark Bond Yield (IT10)FItaly, 3-month Treasury Bill YieldFItaly, 3-month Treasury Bill YieldFSpain, 10-year Government Benchmark Bond Yield (IS10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (IS10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (US10)FUK, 3-month Treasury Bill YieldFGreece, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx S0 Price Index, Euro area (changing composition)FFCAC 40 Price IndexFAC4 40 Price IndexFSdP 500 Composite Price IndexFSdP 100 Price IndexFSdP 500 Composite Price IndexFSdP 500 Composite Price IndexFSdP 500 Composite Price IndexFSdP 500 Composite Price IndexFSdP 500 Co	M 2
European Central Bank deposit rateFEuropean Central Bank lending rateFEuropean Central Bank lending rateFGermany, 10-year Government Benchmark Bond Yield (DE10)FGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Bond Yield (FR10)FFrance, 3-month Treasury Bill YieldFItaly, 10-year Government Benchmark Bond Yield (IT10)FItaly, 3-month Treasury Bill YieldFSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 10-year Government Bond Yield (UK10)FUK, 10-year Government Bond Yield (UK10)FSpread FR10 and DE10FSpread FR10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFCAC 40 Price IndexFCAC 40 Price IndexFSkP 500 Composite Price IndexFSkP 100 Price IndexFSkP 100 Price IndexFSkP 100 Price IndexFSkP 100 Price IndexFSkP 500 Composite Price Index <td>M 2</td>	M 2
European Central Bank lending rateFGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Bond Yield (FR10)FFrance, 3-month Treasury Bill YieldFItaly, 10-year Government Bond Yield (FR10)FItaly, 10-year Government Benchmark Bond Yield (T10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFGreece, 3-month Treasury Bill YieldFUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)FFTSE 100 Price IndexFCAC 40 Price IndexFCAC 40 Price IndexFSder Jon Orposite Price IndexFSder Jon Price IndexFSder Jon Orposite Price IndexFSder Jon Price Index	M 2
Germany, 10-year Government Benchmark Bond Yield (DE10)FGermany, 3-month Treasury Bill YieldFFrance, 10-year Government Bond Yield (FR10)FItaly, 70-year Government Benchmark Bond Yield (T10)FItaly, 3-month Treasury Bill YieldFSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Benchmark Bond Yield (ES10)FGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 10-year Government Bond Yield (UK10)FSpread FR10 and DE10FSpread FR10 and DE10FSpread E110 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFAX 30 Performance Price Index, Euro area (changing composition)FCAC 40 Price IndexFATHEX Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFS&P 100 Price IndexFS&P 100 Price IndexFSWEP Stomposite Price I	M 2
Germany, 3-month Treasury Bul Tield (FR10) F France, 10-year Government Bond Yield (FR10) F Italy, 3-month Treasury Bill Yield F Italy, 3-month Treasury Bill Yield F Spain, 3-month Treasury Bill Yield F Spain, 3-month Treasury Bill Yield (ES10) F Greece, 10-year Government Benchmark Bond Yield (ES10) F Greece, 10-year Government Note Yield (EL10) F Greece, 10-year Government Note Yield (EL10) F Greece, 3-month Treasury Bill Yield F Greece, 10-year Government Note Yield (UK10) F UK, 3-month Treasury Bill Yield F Greece, 3-month Treasury Bill Yield F Spread FR10 and DE10 F Spread FR10 and DE10 F Spread EL10 and DE10 F Euro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition) F Euro Dow Jones Euro Stoxx 70 Price Index, Euro area (changing composition) F TSE 100 Price Index F CAC 40 Price Index F Ster A Derformance Price Index Euro area (changing composition) F Ster A Derformance Price Index F Ster DOW Jones Euro Stoxx 70 Price Index, Euro area (changing composition) F MTHEX Composite Price Index F Ster DOW Jones Euro Stoxx 70 Price Index, Euro area (changing composition) F Ster A Derformance Price Index F Ster DOW Jones Euro Stoxx Price Index F Ster DOW Jones Euro Stoxx Price Index F Ster DOW Jones Euro Index F Ster DOW JONE Price Index F Ster DOW	M 2
France, 3-month Treasury Bill Yield F Italy, 10-year Government Benchmark Bond Yield (IT10) F Italy, 3-month Treasury Bill Yield F Spain, 3-month Treasury Bill Yield F Greece, 10-year Government Benchmark Bond Yield (ES10) F Greece, 10-year Government Note Yield (EL10) F Greece, 3-month Treasury Bill Yield F UK, 10-year Government Bond Yield (UK10) F UK, 3-month Treasury Bill Yield F UK, 10-year Government Bond Yield (UK10) F UK, 3-month Treasury Bill Yield F Spread FL10 and DE10 F Spread FL10 and DE10 F Spread ES10 and DE10 F Spread ES10 and DE10 F Spread ES10 and DE10 F Spread EL10 and DE10 S Spread EL10 S Spre	M 2 M 2
Italy, 10-year Government Benchmark Bond Yield (IT10)FItaly, 3-month Treasury Bill YieldFSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FSpread ES10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFAX 30 Performance Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFSWP 100 Price Index </td <td>M 2</td>	M 2
Italy, 3-month Treasury Bill YieldFSpain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FSpread ES10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)FFTSE 100 Price IndexFAX 30 Performance Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFS&P 100 Price IndexFS&P 100 Price IndexFSWP 100 Pri	M 2
Spain, 10-year Government Benchmark Bond Yield (ES10)FSpain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFSkP 500 Composite Price IndexFSkP 100 Price IndexF<	M 2
Spain, 3-month Treasury Bill YieldFGreece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 70 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFS&P 100 Price IndexFSWP 500 Composite Price Index <t< td=""><td>M 2</td></t<>	M 2
Greece, 10-year Government Note Yield (EL10)FGreece, 3-month Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread FR10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 70 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFNYSE Composite Price IndexFNYESE Composite Price IndexFNUCEY Shore Price IndexFNUCEY Shore Price IndexFSolution Price IndexFSkep 100 Price IndexF	M 2
Checker, 9-Initial Treasury Bill YieldFUK, 10-year Government Bond Yield (UK10)FUK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread IT10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx 70 Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFNYSE Composite Price IndexFNYSE Composite Price IndexFNUCEY Cheven Price IndexFSW F100 Price IndexFSW F200 Composite Price IndexFSW F200 Composite Price IndexFSW F300 Composite Price IndexFSW F300 Price IndexF <t< td=""><td>M 2 M 2</td></t<>	M 2 M 2
UK, 3-month Treasury Bill YieldFSpread FR10 and DE10FSpread IT10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFNYSE Composite Price IndexFNYSE Composite Price IndexFMUCEY Shore Price IndexFSource IndexFSW 100 Price IndexFSwer 500 Composite Pric	M 2
Spread FR10 and DE10FSpread IT10 and DE10FSpread ES10 and DE10FSpread EL10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFS&P 500 Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFSWP 100	M 2
Spread IT10 and DE10FSpread ES10 and DE10FSpread ES10 and DE10FEuro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)FEuro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)FFTSE 100 Price IndexFDAX 30 Performance Price IndexFCAC 40 Price IndexFATHEX Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFNYSE Composite Price IndexFNYSE Composite Price IndexFNYSE Composite Price IndexFStars Davies Ind	M 1
Spread ES10 and DE10     F       Spread EL10 and DE10     F       Euro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)     F       Euro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)     F       FTSE 100 Price Index     F       DAX 30 Performance Price Index     F       CAC 40 Price Index     F       ATHEX Composite Price Index     F       S&P 500 Composite Price Index     F       S&P 100 Price Index     F       NYSE Composite Price Index     F       NUCEY Shore Price Index     F	M 1
Spread EL10 and DE10       F         Euro Dow Jones Euro Stoxx 50 Price Index, Euro area (changing composition)       F         Euro Dow Jones Euro Stoxx Price Index, Euro area (changing composition)       F         FTSE 100 Price Index       F         DAX 30 Performance Price Index       F         CAC 40 Price Index       F         ATHEX Composite Price Index       F         S&P 500 Composite Price Index       F         S&P 500 Composite Price Index       F         NYSE Composite Price Index       F         NUCEY Shows Price Index       F	M 1
Euro Dow Jones Euro Stox 50 Frice Index, Euro area (changing composition)       F         Euro Dow Jones Euro Stox Price Index, Euro area (changing composition)       F         FTSE 100 Price Index       F         DAX 30 Performance Price Index       F         CAC 40 Price Index       F         ATHEX Composite Price Index       F         S&P 500 Composite Price Index       F         S&P 100 Price Index       F         NYSE Composite Price Index       F         MUCEY Shows Price Index       F	M 1
FISE 100 Price Index File Index File Index, File Index, File Index, File Index File Inde	M 3
DAX 30 Performance Price Index F CAC 40 Price Index F ATHEX Composite Price Index F S&P 500 Composite Price Index F S&P 100 Price Index F NYSE Composite Price Index F	M 3
CAC 40 Price IndexFATHEX Composite Price IndexFS&P 500 Composite Price IndexFS&P 100 Price IndexFNYSE Composite Price IndexFNVSE Composite Price IndexF	М 3
ATHEX Composite Price Index     F       S&P 500 Composite Price Index     F       S&P 100 Price Index     F       NYSE Composite Price Index     F       MUCEY Share Price Index     F	М 3
S&P 500 Composite Price Index     F       S&P 100 Price Index     F       NYSE Composite Price Index     F       MUCEY Share Price Index     F	M 3
S&P 100 Price Index F NYSE Composite Price Index F	M 3
N ISE Composite Frite index F	M 3 M 2
	M 3
Nikkei 225 Stock Average Index F	M 3
CBOE S&P 500 Volatility Index F	M 1
EURO STOXX 50 Price Index, historical volatility (based on historical data of EURO STOXX 50 Price Index) F	1 1/1
S&P500 Index, historical volatility (based on historical data of S&P500 Index) F	WI 1
Exchange rate, euro against the British pound F	M 1
Exchange rate, euro against the US dollar E	M 1 M 3 M 2
Exchange rate, euro against the Chinese renminbi vuan	M 1 M 3 M 3 M 3
EA, Harmonised Index of Consumer Prices (HICP), (2015=100)	M 1 M 3 M 3 M 3 M 3 M 3 M 3
EU, Harmonised Index of Consumer Prices (HICP), (2015=100)	M 1 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3
UK, Harmonised Index of Consumer Prices (HICP), (2015=100) F	M 1 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3
Russia, Consumer Price Index F	M 1 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3
Brent Crude Oil (€) F	M 1 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3
west rexas intermediate Oil Price (t/ barrel)     F       Crude Oil Futures (c)     Futures Contracts	M 1 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3 M 3

### TABLE A2 (continued)

Variable description		Details		
variable description	А	В	С	
Gold Bullion Price - New York (€/Ounce)	F	Μ	3	Ì
Silver Cash Price (€/Ounce)	F	Μ	3	
Wheat Cash Price ( $\epsilon$ /Bushel)	F	Μ	3	
Euro area 17 (fixed composition) ECB Commodity Price index, Euro denominated, use-weighted, Food	F	Μ	3	
Euro area 17 (fixed composition) ECB Commodity Price index, Euro denominated, use-weighted, Non-food	F	Μ	3	
Euro area 17 (fixed composition) ECB Commodity Price index, Euro denominated, use-weighted, Total non-energy commodity	F	Μ	3	
EA, HICP inflation expectations, current calendar year (ECB Survey of Professional Forecasters)	F	Q	1	
EA, HICP inflation expectations, next calendar year (ECB Survey of Professional Forecasters)	F	Q	1	
Global Economic Policy Uncertainty Index	F	Μ	1	
EU, Economic Sentiment Indicator	F	Μ	2	
EA, Economic Sentiment Indicator	F	Μ	2	
UK, Economic Sentiment Indicator	F	Μ	2	
Greece, Economic Sentiment Indicator	F	Μ	2	
EU, Employment Expectations Indicator	F	Μ	2	
EA, Employment Expectations Indicator	F	Μ	2	
EU, Services Confidence Indicator	F	Μ	2	
EU, Industry Confidence Indicator	F	Μ	2	
EU, Retail Trade Confidence Indicator	F	Μ	2	
EU, Construction Confidence Indicator	F	Μ	2	
EU, Consumer Confidence Indicator	F	Μ	2	
EA, Services Confidence Indicator	F	Μ	2	
EA, Retail Trade Confidence Indicator	F	Μ	2	
EA, Construction Confidence Indicator	F	Μ	2	
EA, Industry Expectations Indicator	F	Μ	2	
EA. Consumer Confidence Indicator	F	М	2	

Notes: The column "Details" is read as follows: in column A the symbols "D" and "F" denote domestic and foreign/international indicators, respectively; in column B the symbols "M" and "Q" denote monthly and quarterly data frequency, respectively; in column C the transformation codes 1, 2 and 3 represent a series in levels, first difference of levels, and first difference of the logarithm of levels, respectively.

# TABLE A3 Percentages of selected predictors by type, and cut-off frequencies, production-side components

	1			1					
Horizon (quarters)		1			4			8	
		0			0			0	
Predictor selection method	test	ASS	ARS	test	ASS	ARS	test	ASS	ARS
	ц	Ľ	L	ц	Γ	Ę	ц	Γ	Ľ
AGR Types of predictors									
Domestic	71	65	52	36	29	25	40	35	25
Foreign/international	29	35	48	64	71	75	60	65	76
By category									
Real activity and labour market	5	25	19	36	29	30	20	25	40
Financial	14	30	33	14	38	30	50	50	35
Prices	19	10	19	32	14	20	20	10	10
Business and consumer surveys	62	35	29	18	19	20	10	15	15
Cut-off frequency (%)	38	29	28	26	20	23	35	23	18
IND Terra of the distant									
Types of predictors	62	76	65	45	19	50	55	12	45
Foreign/international	38	24	35	40	40 52	50	45	43 57	40 55
By category	50	24	55	55	52	50	-10	57	55
Real activity and labour market	38	33	25	5	10	0	25	14	5
Financial	10	38	35	45	52	45	40	57	60
Prices	38	5	15	30	19	30	5	10	10
Business and consumer surveys	14	24	25	20	19	25	30	19	25
Cut-off frequency (%)	33	29	31	38	29	29	34	28	29
CON									
Types of predictors									
Domestic	45	70	76	50	70	62	45	45	48
Foreign/international	55	30	24	50	30	38	55	55	52
Boal activity and labour market	45	30	13	35	55	38	20	35	20
Financial	25	30	24	40	35	48	50	50	29 52
Prices	5	5	5	10	5	5	15	10	10
Business and consumer surveys	25	35	29	15	5	10	15	5	10
Cut-off frequency (%)	39	28	24	36	29	28	40	28	23
TRA									
Types of predictors									
Domestic	40	65	38	15	30	30	35	43	40
Foreign/international	60	35	62	85	70	70	65	57	60
By category	10	10	•	0	4.0	40	40	45	•
Real activity and labour market	10	40	29	0	10	10	10	17	20
Pricos	40	23	55 10	55	40 30	40 25	15	22	40 20
Business and consumer surveys	50	30	29	40	20	25	20	26	20
Cut-off frequency (%)	40	33	36	46	25	23	45	32	34
INF									
Types of predictors									
Domestic	67	71	57	38	55	41	35	48	35
Foreign/international	33	29	43	62	45	59	65	52	65
By category									
Real activity and labour market	10	19	5	10	5	9	20	13	30
Financial	19	29	52	43	35	50	40	39	40
Prices	14	14	10	10	15	14	20	1/	20
Cut-off frequency (%)	35	36 26	31	58 41	45 28	27	20 45	25	26
Types of predictors	35	20	51	71	20	20	-10	25	20
FIN									
Types of predictors									
Domestic	40	61	60	30	48	35	20	48	33
Foreign/international	60	39	40	70	52	65	80	52	67
By category									
Real activity and labour market	45	43	50	10	29	20	10	30	24
Financial	35	17	20	60	24	35	80	39	38
Prices	15	22	15	0	10	10	0	13	10
Cut-off frequency (%)	3	25	15	30	28 28		10	28	29
$\mathcal{L}_{u}$	40	23	<u>∠4</u>	39	∠0	23	40	20	27

# TABLE A3 (continued)

Horizon (quarters)		1			4			8	
Predictor selection method	F-test	LASSO	LARS	F-test	LASSO	LARS	F-test	LASSO	LARS
REA									
Types of predictors									
Domestic	30	52	45	18	36	30	48	60	52
Foreign/international	70	48	55	82	64	70	52	40	48
By category									
Real activity and labour market	10	10	10	18	14	15	29	20	22
Financial	65	38	45	50	41	40	48	40	39
Prices	15	24	20	18	23	30	5	15	9
Business and consumer surveys	10	29	25	14	23	15	19	25	30
Cut-off frequency (%)	33	24	29	35	26	26	29	22	17
PRO									
Types of predictors	25	50	F7	20	20	25	10	20	20
Domestic E-minuter (internetional)	35	50	57	20	20	25	19	30	29
Foreign/international	65	50	43	80	80	75	81	70	/1
Bool activity and labour market	20	20	20	20	15	20	14	20	24
Financial	30	20	29	20 50	50	20	14 76	20 45	2 <del>4</del> 13
Prices	0	0	0	5	10	10	0	-10	10
Business and consumer surveys	40	45	38	25	25	20	10	30	24
Cut-off frequency (%)	40	31	24	36	32	25	43	26	18
PFH	10	01	21	50	02	20	10	20	10
Types of predictors									
Domestic	48	52	50	38	45	35	60	55	50
Foreign/international	52	48	50	62	55	65	40	45	50
By category									
Real activity and labour market	13	10	10	33	5	25	50	20	20
Financial	17	19	35	14	45	35	40	40	35
Prices	0	10	5	5	0	10	0	5	10
Business and consumer surveys	70	62	50	48	50	30	10	35	35
Cut-off frequency (%)	31	28	28	32	29	26	38	28	25
OTH									
Types of predictors									
Domestic	40	55	45	25	40	30	33	38	25
Foreign/international	60	45	55	75	60	70	66	62	75
By category									
Real activity and labour market	15	15	15	15	20	15	33	19	25
Financial	25	30	25	40	50	45	43	48	50
Prices	35	30	40	30	15	20	14	19	10
Cut off froguen gy (%)	42	23	20	13	13	20	10	14	13
Cut-on frequency (%)	42	32	33	38	28	30	42	22	26
Types of predictors									
Domostic	50	75	80	20	35	20	35	40	30
Foreign/international	50	25	20	29 71	65	29 71	65	40 60	70
By category	50	20	20	/1	00	/1	00	00	70
Real activity and labour market	35	30	50	5	25	24	5	5	10
Financial	20	20	20	38	20	29	45	40	45
Prices	0	5	0	19	20	19	20	15	10
Business and consumer surveys	45	45	30	38	35	29	30	40	35
Cut-off frequency (%)	36	31	29	38	33	30	45	29	32

Note: The cut-off frequency shows the lowest frequency with which an indicator should be chosen across sample iterations by a given method in order to be included in the set of selected predictors.

Horizon (quarters)			1			4	4			8	8	
Predictor selection method	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors
AGR												
Benchmark AR(1): RMSFE		2.	49			8.	55			9.	13	
Scheme (a): each <i>t</i>												
FAR - mean	1.21	1.28	1.23	1.09	1.29	1.28	1.27	1.15	1.20	1.44	1.42	1.14
FAR - DMSFE	1.21	1.29	1.23	1.09	1.30	1.27	1.27	1.14	1.20	1.44	1.42	1.14
ADL - mean	1.05	1.05	1.05	1.02	1.17	1.08	1.09	1.07	1.10	1.16	1.18	1.05
ADL - DMSFE	1.08	1.05	1.03	1.02	1.12	1.11	1.13	1.08	1.18	1.13	1.20	1.05
Scheme (b): all <i>t</i>												
FAR - mean	1.05	1.12	1.10	1.09	1.14	0.96	1.09	1.15	1.24	1.11	1.02	1.14
FAR - DMSFE	1.06	1.10	1.10	1.09	1.12	0.96	1.08	1.14	1.22	1.13	1.02	1.14
ADL - mean	0.99	1.00	1.01	1.02	1.04	1.04	1.03	1.07	1.01	1.03	1.06	1.05
ADL - DMSFE	0.99	1.01	1.01	1.02	1.03	1.04	1.03	1.08	1.04	1.06	1.08	1.05
IND												
Benchmark AR(1): RMSFE		1.	87			6.	13			8.	21	
Scheme (a): each t	4.00	4.00	0.07		4.04	0.04	0.04		4.40	4.07		
FAR - mean	1.00	1.03	0.97	0.88	1.04	0.94	0.94	0.88	1.13	1.06	1.14	1.12
FAR - DMSFE	1.00	1.01	0.97	0.88	1.04	0.94	0.94	0.87	1.13	1.06	1.14	1.13
ADL - mean	0.91	0.94	0.93	0.95	1.02	1.02	1.01	0.99	1.01	0.99	0.95	1.04
ADL - DMSFE	0.91	0.94	0.93	0.95	1.02	1.02	1.01	0.99	0.95	0.95	0.00	1.03
Scheme (b): all t	0.87	0.02	0.84	0.88	0.06	0.86	0.04	0.88	0.00	1.00	1.06	1 1 2
FAR - mean FAR DMSEE	0.87	0.93	0.04	0.88	0.90	0.86	0.94	0.87	1.00	1.00	1.00	1.12
ADI - mean	0.07	0.93	0.05	0.00	0.97	0.00	0.94	0.07	0.93	0.98	0.98	1.15
ADL - DMSFE	0.90	0.95	0.95	0.95	0.87	0.91	0.95	0.99	0.95	0.90	0.90	1.04
	0.09	0.71	0.71	0.95	0.07	0.90	0.72	0.77	0.00	0.71	0.70	1.05
Benchmark AR(1): RMSFE		5	09			15	57			18	48	
Scheme (a): each $t$		0.				10				10		
FAR - mean	1.21	1.15	1.16	0.91	1.02	1.06	1.03	1.04	0.95	0.94	1.04	1.13
FAR - DMSFE	1.20	1.17	1.13	0.90	1.02	1.06	1.03	1.05	0.95	0.94	1.04	1.13
ADL - mean	0.91	0.93	0.93	0.91	0.94	0.89	0.91	0.88	1.02	0.99	0.99	1.00
ADL - DMSFE	0.94	1.00	0.99	0.93	0.95	0.97	0.91	0.88	1.03	0.98	1.03	1.00
Scheme (b): all <i>t</i>												
FAR - mean	0.88	0.77	0.85	0.91	1.09	0.74	0.67	1.04	1.06	0.73	0.68	1.13
FAR - DMSFE	0.86	0.78	0.85	0.90	1.09	0.74	0.67	1.05	1.08	0.73	0.68	1.13
ADL - mean	0.86	0.89	0.88	0.91	0.84	0.83	0.86	0.88	0.96	0.96	0.96	1.00
ADL - DMSFE	0.89	0.90	0.89	0.93	0.84	0.84	0.85	0.88	0.93	0.94	0.94	1.00
TRA												
Benchmark AR(1): RMSFE	1.60				5.	21			6.	04		
Scheme (a): each <i>t</i>												
FAR - mean	0.99	0.90	0.92	0.99	0.87	0.77	0.71	1.01	1.00	1.07	1.10	1.15
FAR - DMSFE	0.98	0.90	0.91	0.99	0.87	0.77	0.71	1.01	1.00	1.07	1.11	1.15
ADL - mean	0.89	0.88	0.86	0.93	0.86	0.81	0.79	0.93	1.01	0.97	0.98	0.98
ADL - DMSFE	0.87	0.88	0.85	0.93	0.90	0.83	0.81	0.92	1.16	0.98	1.01	0.98
Scheme (b): all t	0.00	0.70	0.90	0.00	0.90	0.69	0.47	1.01	0.90	0.02	0.97	1.15
FAR - MEAN	0.90	0.79	0.80	0.99	0.89	0.68	0.67	1.01	0.89	0.93	0.86	1.15
ADI moon	0.91	0.78	0.80	0.99	0.89	0.09	0.67	1.01	0.89	0.93	0.86	1.15
	0.86	0.84	0.81	0.93	0.85	0.78	0.74	0.93	0.84	0.88	0.87	0.98

TABLE A4
Relative RMSFE (benchmark AR), production-side components

### TABLE A4 (continued)

Horizon (quarters)			1			4		8						
Predictor selection method	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors		
INF Bonchmort AP(1): PMSEE		5	24		11.00				11.10					
Scheme (a): each $t$		.34	11.29				11.15							
FAR - mean	1.13	1.14	1.14	1.05	1.17	1.40	1.23	1 24	1.13	1.36	1.39	1.00		
FAR - DMSFE	1.14	1.14	1.12	1.05	1.16	1.40	1.23	1.18	1.13	1.36	1.39	1.00		
ADL - mean	1.08	1.06	1.07	1.05	1.04	1.01	1.00	1.01	1.06	1.03	1.08	1.00		
ADL - DMSFE	1.06	1.08	1.07	1.04	1.03	1.04	1.01	1.02	1.10	1.13	1.32	1.01		
Scheme (b): all <i>t</i>														
FAR - mean	1.09	1.08	1.01	1.05	1.09	0.90	0.93	1.24	1.26	0.91	0.94	1.00		
FAR - DMSFE	1.09	1.05	1.01	1.05	1.08	0.91	0.93	1.18	1.25	0.91	0.94	1.00		
ADL - mean	1.04	1.04	1.04	1.05	0.93	0.96	0.93	1.01	0.98	0.98	0.98	1.00		
ADL - DMSFE	1.04	1.04	1.03	1.04	0.94	0.98	0.93	1.02	0.98	0.99	1.00	1.01		
FIN														
Benchmark AR(1): RMSFE			9.	05			10	.31						
Scheme (a): each <i>t</i>														
FAR - mean	1.80	1.29	2.19	1.08	1.26	1.18	1.11	1.21	1.22	1.05	1.11	1.28		
FAR - DMSFE	1.72	1.31	2.10	1.13	1.26	1.18	1.11	1.21	1.23	1.05	1.11	1.27		
ADL - mean	0.98	0.99	0.99	0.97	1.03	0.98	0.97	1.00	1.02	1.03	1.06	1.00		
ADL - DMSFE	0.98	1.06	0.99	0.94	1.07	1.02	0.98	1.00	1.00	0.98	1.04	1.00		
Scheme (b): all <i>t</i>	1.02			4.00	1.00	4.40	1.00	4.04	4.00	4.40		4.00		
FAR - mean	1.03	1.11	1.11	1.08	1.23	1.12	1.02	1.21	1.20	1.10	1.14	1.28		
FAR - DMSFE	1.04	1.11	1.10	1.13	1.20	1.12	1.02	1.21	1.20	1.10	1.13	1.27		
ADL - mean	0.94	0.95	0.94	0.97	0.99	0.98	0.95	1.00	0.97	1.00	0.99	1.00		
ADL - DWSFE	0.93	0.92	0.90	0.94	0.96	0.94	0.92	1.00	0.96	0.98	0.96	1.00		
REA Bonchmark AR(1): RMSEE		2	16			2.06				4 60				
Scheme (a): each $t$		۷.	.40		5.96				<b>1.00</b>					
FAR - mean	1.05	0.98	0.95	1.00	1.10	1.01	1.05	1.05	1.11	1.21	1.16	1.05		
FAR - DMSFE	1.05	0.98	0.95	1.00	1.10	1.01	1.05	1.00	1.10	1.21	1.16	1.05		
ADL - mean	1.02	1.02	1.02	1.04	1.02	1.01	1.01	1.06	1.06	1.04	1.03	1.03		
ADL - DMSFE	1.02	1.03	1.02	1.04	1.01	1.02	1.03	1.05	1.04	1.02	1.02	1.03		
Scheme (b): all <i>t</i>														
FAR - mean	1.01	0.98	0.99	1.00	1.01	0.79	0.84	1.05	1.00	1.01	0.92	1.05		
FAR - DMSFE	1.01	0.98	0.99	1.01	1.00	0.80	0.84	1.04	1.00	1.01	0.92	1.05		
ADL - mean	1.02	1.02	1.02	1.04	1.02	1.02	1.02	1.06	1.02	1.02	1.03	1.03		
ADL - DMSFE	1.02	1.02	1.02	1.04	1.02	1.02	1.02	1.05	1.02	1.02	1.03	1.03		
PRO														
Benchmark AR(1): RMSFE	1.43				4.48				5.15					
Scheme (a): each <i>t</i>														
FAR - mean	1.06	1.17	1.24	1.01	1.34	1.31	1.36	1.27	0.98	1.05	1.11	1.26		
FAR - DMSFE	1.06	1.14	1.21	1.01	1.34	1.31	1.36	1.27	0.98	1.06	1.11	1.26		
ADL - mean	0.98	0.95	0.96	0.97	1.04	0.97	0.99	0.98	0.99	0.97	0.97	0.99		
ADL - DMSFE	1.00	0.96	0.94	0.96	1.07	1.09	1.09	0.97	1.03	0.98	0.98	1.00		
Scheme (b): all t	1.07	1.00	1.04	1.01	1 10	1.00	1.00	1.07	1.04	0.07	0.00	1.0/		
FAK - mean	1.07	1.03	1.04	1.01	1.12	1.08	1.08	1.27	1.34	0.86	0.93	1.26		
ADI moan	1.07	1.05	1.04	1.01	1.11	1.08	1.09	1.27	1.33	0.80	0.93	1.20		
	0.94	0.94	0.92	0.97	0.90	0.91	0.92	0.90	0.95	0.92	0.93	1.00		
ADL - mean ADL - DMSFE	0.94 0.92	0.94 0.92	0.92 0.92	0.97 0.96	0.90 0.89	<b>0.91</b> 0.92	0.92 0.92	0.98 0.97	0.93 0.96	<b>0.92</b> 0.95	<b>0.93</b> 0.96	0.99 1.00		

### TABLE A4 (continued)

Horizon (quarters)			1		4				8					
Predictor selection method	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors	F-test	LASSO	LARS	All predictors		
PEH														
Benchmark AR(1): RMSFE		0.	71		2.27				2.93					
Scheme (a): each t	110	1.00	1.00	4.04	1.15	1 10	1 01	1.00	1.00	1 11	1 1 2	0.00		
FAR - mean	1.10	1.08	1.09	1.01	1.15	1.18	1.21	1.02	1.08	1.11	1.15	0.89		
FAR - DMSFE	1.16	1.08	1.08	1.01	1.15	1.17	1.21	1.03	1.08	1.11	1.13	0.89		
ADL - mean	1.02	1.00	0.98	1.00	1.05	0.96	1.04	1.03	0.97	0.96	0.00	1.06		
Scheme (b): all $t$	1.02	1.00	0.97	1.00	1.11	0.90	1.04	1.05	0.95	0.90	0.22	1.03		
FAR - mean	1.00	0.98	1.05	1.01	0.81	0.64	1 13	1.02	0.92	0.92	1.02	0.89		
FAR - DMSFF	1.00	0.98	1.05	1.01	0.80	0.64	1.15	1.02	0.92	0.92	1.02	0.89		
ADL - mean	0.94	0.98	0.96	1.01	0.89	0.98	0.98	1.03	0.92	1.02	1.02	1.06		
ADL - DMSFE	0.94	0.98	0.95	1.00	0.88	0.98	0.97	1.03	0.87	0.99	0.98	1.00		
ОТН	017 -		0.00											
Benchmark AR(1): RMSFE		1.	91			7.32				8.55				
Scheme (a): each <i>t</i>														
FAR - mean	0.98	0.96	1.09	1.11	1.13	1.12	1.17	1.01	1.20	1.26	1.13	1.12		
FAR - DMSFE	0.96	0.95	1.11	1.06	1.13	1.12	1.18	1.02	1.19	1.26	1.12	1.12		
ADL - mean	1.02	0.99	1.00	1.02	1.02	1.04	1.04	1.05	1.03	1.03	1.00	0.99		
ADL - DMSFE	1.03	1.00	1.01	1.01	1.00	1.05	1.05	1.04	1.03	1.12	1.08	0.99		
Scheme (b): all <i>t</i>														
FAR - mean	0.98	1.05	1.01	1.11	1.16	0.89	0.85	1.01	1.16	1.20	1.25	1.12		
FAR - DMSFE	0.96	1.04	0.97	1.06	1.16	0.90	0.85	1.02	1.16	1.20	1.25	1.12		
ADL - mean	0.97	0.98	0.97	1.02	0.99	1.03	1.02	1.05	1.02	1.00	1.01	0.99		
ADL - DMSFE	0.95	0.97	0.96	1.01	0.98	1.02	1.00	1.04	1.01	1.00	1.00	0.99		
TAX														
Benchmark AR(1): RMSFE	1.33				4.02				4.82					
Scheme (a): each <i>t</i>														
FAR - mean	0.85	0.91	0.76	0.83	0.73	0.70	0.66	0.81	1.01	1.07	1.06	1.01		
FAR - DMSFE	0.87	0.94	0.77	0.81	0.72	0.69	0.66	0.80	1.01	1.07	1.06	1.01		
ADL - mean	0.77	0.83	0.82	0.85	0.87	0.81	0.84	0.93	1.00	0.98	0.97	1.00		
ADL - DMSFE	0.74	0.78	0.69	0.80	0.77	0.86	0.89	0.92	1.00	1.08	0.97	1.00		
Scheme (b): all <i>t</i>														
FAR - mean	0.72	0.72	0.94	0.83	0.50	0.53	0.50	0.81	0.96	0.92	0.89	1.01		
FAR - DMSFE	0.72	0.71	0.88	0.81	0.50	0.53	0.49	0.80	0.95	0.92	0.89	1.01		
ADL - mean	0.73	0.79	0.78	0.85	0.75	0.80	0.80	0.93	0.87	0.89	0.88	1.00		
ADL - DMSFE	0.68	0.70	0.70	0.80	0.75	0.78	0.78	0.92	0.88	0.88	0.88	1.00		

See notes to Table 3.

#### References

Baffigi, A., Golinelli, R. and Parigi, G., (2004), "Bridge models to forecast the euro area GDP", *International Journal of Forecasting*, 20(3): 447-460.

Bai, J., and Ng, S., (2002), "Determining the number of factors in approximate factor models", *Econometrica*, 70(1): 191-221.

Bai, J., and Ng, S., (2006), "Confidence intervals for diffusion index forecasts and inference for factor-augmented regressions", *Econometrica*, 74(4): 1133-1150.

Bai, J., and Ng, S., (2008), "Forecasting economic time series using targeted predictors", *Journal of Econometrics*, 146(2): 304-317.

Banbura, M. and Runstler, G., (2011), "A look into the factor model black box: publication lags and the role of hard and soft data in forecasting GDP", *International Journal of Forecasting*, 27(2): 333-346.

Boivin, J., and Ng, S., (2006), "Are more data always better for factor analysis?", *Journal of Econometrics*, 132(1): 169-194.

Bulligan, G., Marcellino, M., and Venditti, F., (2015), "Forecasting economic activity with targeted predictors", *International Journal of Forecasting*, 31(1): 188-206.

Caggiano, G., Kapetanios, G. and Labhard, V., (2011), "Are more data always better for factor analysis? Results for the euro area, the six largest euro area countries and the UK", *Journal of Forecasting*, 30(8): 736-752.

Diebold, F. X. and Mariano, R. S. (1995), "Comparing predictive accuracy", *Journal of Business and Economic Statistics*, 13(3): 134-144.

Efron, B., Hastie, T., Johnstone, I., and Tibshirani, R., (2004), "Least angle regression", *Annals of Statistics*, 32(2): 407-499.

Eurostat (2013), Handbook on quarterly national accounts.

Foroni, C., and Marcellino, M., (2014), "A comparison of mixed frequency approaches for nowcasting Euro area macroeconomic aggregates", *International Journal of Forecasting*, 30(3): 554-568.

Giannone, D., Reichlin, L. and Small, D., (2008), "Nowcasting: The real-time informational content of macroeconomic data", *Journal of Monetary Economics*, 55(4): 665-676.

Girardi, A., Golinelli, R., and Pappalardo, C., (2017), "The role of indicator selection in nowcasting euro-area GDP in pseudo-real time", *Empirical Economics*, 53(1): 79-99.

Harvey, D., Leybourne, S. and Newbold, P. (1997), "Testing the equality of prediction mean squared errors", *International Journal of Forecasting*, 13(2): 281-291.

Kim, H. H., and Swanson, N. R., (2014), "Forecasting financial and macroeconomic variables using data reduction methods: New empirical evidence", *Journal of Econometrics*, 178: 352-367.

Kim, H. H. and Swanson, N. R., (2018), "Mining big data using parsimonious factor, machine learning, variable selection and shrinkage methods", *International Journal of Forecasting*, 34(2): 339-354.

Kitchen, J., and Monaco, R., (2003), "Real-time forecasting in practice: The U.S. treasury staff's real-time GDP forecast system", *Business Economics*, 4(38): 10-19.

Jokubaitis, S., Celov, D. and Leipus, R., (2021), "Sparse structures with LASSO through principal components: Forecasting GDP components in the short-run", *International Journal of Forecasting*, 37(2): 759-776.

Panagiotelis, A., Athanasopoulos, G., Hyndman, R. J., Jiang, B., and Vahid, F., (2019), "Macroeconomic forecasting for Australia using a large number of predictors", *International Journal of Forecasting*, 35(2): 616-633.

Pashourtidou, N., Papamichael, C., and Karagiannakis, C., (2018), "Forecasting economic activity in sectors of the Cypriot economy", *Cyprus Economic Policy Review*, 12(2): 24-66.

Schumacher, C., (2010), "Factor forecasting using international targeted predictors: The case of German GDP", *Economics Letters*, 107(2): 95-98.

Stock, J. H. and Watson, M. W., (2002), "Forecasting using principal components from a large number of predictors", *Journal of the American Statistical Association*, 97(460): 1167-1179.

Stock, J. H. and Watson, M. W., (2004), "Combination forecasts of output growth in a seven-country data set", *Journal of Forecasting*, 23:405-430.

Stock, J. H. and Watson, M. W., (2006), "Forecasting with many predictors", *Handbook of Economic Forecasting*, 1: 515-554.